DIRECT SECANT ESTIMATION OF LIMIT BEHAVIOUR OF FRAMED STRUCTURES

XIAO YU







DIRECT SECANT ESTIMATION OF LIMIT BEHAVIOUR OF FRAMED STRUCTURES

by

Xiao Yu

A thesis submitted to the

School of Graduate Studies

in partial fulfillment of the

requirements for the degree of

Master of Engineering

Faculty of Engineering and Applied Science

Memorial University of Newfoundland

January 2010

St. John's

Newfoundland

Abstract

A direct secant method used to predict the plastic limit load of framed structures is reesented in this thesis. Instead of using classical techniques, it utilizes two or more purely elastic analyses to predict the limit load. Secant rigidity of structures is modified based on the result of the first purely elastic analysis. Iterative reanalyses are performed until convergence is reached. By selecting the peak bending moment, rotential plastic hinges are listed. The result can be used to predict the plastic limit load as well as the collapse mechanism. The limit load calculated by the direct secant method is compared to the solution of other traditional analyses, where applicable. Generally, the direct secant method is an attractive alternative for evaluating the limit load of framed structures. The results are a significant improvement over traditional methods which are illustrated in the thesis. Similar idea is applied on frame stability analysis. After the first purely elastic analysis, two preliminary and empirical methods for analyzing stability are suggested. Factor C and factor ß are investigated to evaluate the critical load. This thesis investigates large deflection by using similar idea inspired from the direct secant method used for analyzing frame stability. Factor n is investigated and used to analyzing the large deflection

The method is executed by ANSYS software using APDL routines. The problems solved include: Portal frames, Two-bay Single-torry frame, Two-bay Two-torry frame as well as Multistorey Frame Subject to Concentrated and Distributed Loads. The results from the above analyses compare with other traditional methods closely.

1

and the error is no more than 5%, thus demonstrating the usefulness of the direct

secant method.

Acknowledgment

I with a capters ny deepet akawahdgement and patritude to my supervise. Dr Schah Madhwa Rao Athui, 1 sincerly dunk him fer his my supervise. The selection of t

Appreciation is extended to Dr Hesham Marzouk and Dr Glyn Geroge for their excellent course teaching. Gratitude goes to faculty of engineering and graduate student office for their financial support.

Thanks to my parents for their personal life care and spiritual support. They are the most close persons I have in the world. I show my special thankfulment to my wife Dan Dan. She gave me the most powerful spiritual support and countless resourcement. She is beautiful. How her.

Thankfulness is sent to all my friends at Memorial University for the help and happiness they brought to me. The time I spent with them will cave into my memory. Thanks WenShu Bodhisattva.

-

Table of Contents

Abstract	I
Acknowledgment	
Table of Contents	IV
List of Figures	X
List of Tables	XIV
List of Symbols	XV
Chapter 1 Introduction	
1.1 General	1
1.2 Objectivities	4
1.3 Organization of the Thesis	
Chapter 2 Literature Review	6
2.1 Buckling and Structural Stability	
2.1.1 Stable Equilibrium	
2.1.2 Unstable Equilibrium	
2.1.3 Neutral Equilibrium	
2.2 Principle of Stationary Potential Energy	
2.3 Basic Approaches for Critical Load	9
2.4 Upper Bound Theorem and Lower Bound Theorem	
2.4.1 Upper bound Theorem	
2.4.2 Lower Bound Theorem	
2.5 Critical Load of Columns with Various Supports	
2.5.1 Ideal Column	
2.5.2 Enler Emution	11

		2.5.3	Critical Load of Columns with Various Supports	
	2.6	Lir	nit Load and Limit State	
		2.6.1	Limit Load	
		2.6.2	Limit State	
	2.7	Ela	stic Perfectly- plastic Model	
		2.7.1	Stress-strain Relation of Mild Steel	
		2.7.2	Elastic Perfectly- plastic Model	
	2.8	Fai	lure Criterion	
		2.8.1	Maximum Principle Stress Yield Stress Criterion	
		2.8.2	Von Mises Yield Criterion	
	2.9	No	nlinear Analysis of Structures	
	2.10	0 Lir	nit Load Prediction	
		2.10.1	Reference Stress Method	
		2.10.2	Partial Elastic Modulus Modification	
		2.10.3	Gloss R-Node Method	
		2.10.4	Gloss Method	
		2.10.5	R-Node Method	
		2.10.6	The m-a Method	
		2.10.7	Nesting Surface Theorem	
		2.10.8	Extended Lower Bounded Theorem	
CI	apte	r3 P	lastic Limit Load Estimation	
	3.1	Intro	duction	
	3.2	Plas	tic Analysis Review	
		3.2.1	Mechanisms of Failure for Frames	
		3.2.2	Plastic Design of Structures	

3.3	Alt	ernative Methods	
3.4	Ins	pirations from Plastic Hinge and R-Node Analysis	
3.5	Th	e Direct Secant Method	
3.6	Pla	stic Limit Load of Non-uniform Structures	
3.7	Co	mparison between Direct Secant Method and R-Node Method	62
	3.7.1	Modification	62
	3.7.2	Element Type	63
	3.7.3	Failure mode and yield criterion	63
	3.7.4	Limit load	
3.8	Ter	t Cases and Examples	64
	3.8.1	Finite Element Modeling	
	3.8.2	The Description of the Test Cases and Examples	
Chapte	r4 E	stimations for Frame Stability	
4.1	Int	roduction	
4.2	Ob	servations from Elastic Buckling Theory	
4.3	Ins	piration from Robust Plastic Limit Load Analysis	
4.4	Ro	bust "Secant" Analysis Trials for Elastic Buckling	
	4.4.1	Method 1 for Critical Loads	
	4.4.2	Method 2 for Critical Loads	
	4.4.3	Comparison Between Method 1 and Method 2	
4.5	Re	sults	
Chapte	r 5 Est	imation of Large Deflections in Beams	
5.1	Introdu	ection	118
5.2	Euler-	Bemoulli Beam	119
	5.2.1	Linear Theory	119

	5.2.2	Modified Euler-Bernoulli Equation
5.3	Sei	ope of the Analysis
5.4	Tb	e Direct Secant Technique
	5.4.1	Cantilever Beam
	5.4.2	Simply Supported Beam with a Roller Support
	5.4.3	Simply Supported Beam without Roller Support
	5.4.4	Portal Frame with Lateral Load
Chapte	r 6 Co	aclusions
6.1	Int	roduction
6.2	Su	mmary
6.3	Co	nclusions
	6.3.1	Plastic Limit Loads Estimation
	6.3.2	Buckling
	6.3.3	Large Deflection Analysis
6.4	Re	commendations for Further Work
REFER	RENCE	145 IS
APPEN	DICE	s152
APPEN	DIX A	
A.1	.1 A I	UNIFORM PORTAL FRAME SUBJECT TO A LATERAL FORCE
AN	DAV	ERTICAL FORCE
A.1	1.2 A	TWO BAY AND ONE STOREY FRAME SUBJECT TO TWO
VE	RTICA	L FORCES AND ONE LATERAL FORCE
A.1	.3 A U	NIFORM FRAME SUBJECT TO THREE VERTICAL FORCES AND
ON	E LAT	ERAL FORCE
A.1	14 A	UNIFORM BIG FRAME SUBJECT TO UNIFORMLY LATERAL

	DISTRIBUTED FORCES AND VERTICAL CONCENTRATED FORCES 157	
	A.1.5 A NON-UNIFORM PORTAL FRAME SUBJECT TO A LATERAL	
	FORCE AND A VERTICAL FORCE	
	A.2.1 A UNIFORM PORTAL FRAME SUBJECT TO THREE VERTICAL	
	FORCES ON THE BEAM	
	A.2.2 A UNIFORM PORTAL FRAME SUBJECT TO A VERTICAL FORCE	
	AND A LATERAL FORCE	
	A.3.1 A CANTILEVER BEAM SUBJECT TO A CONCENTRATED FORCE AT	
	THE FREE END	
	A.3.2 A SIMPLY SUPPORTED BEAM WITH ONE ROLLER SUPPORT	
	SUBJECT TO A CONCENTRATED FORCE IN THE MIDDLE	
	A.3.3 A SIMPLY SUPPORTED BEAM WITHOUT ROLLER SUPPORT	
	SUBJECT TO A CONCENTRATED FORCE IN THE MIDDLE	
J	PENDIX B	
	B.1 MACROI FOR MODIFYING SECANT RIGIDITY IN PLASTIC	
	ANALYSIS	
	B.2 MACRO2 FOR MODIFYING SECANT RIGIDITY IN BUCKLING	
	ANALYSIS	
	B.3 MACRO3 FOR MODIFYING SECANT RIGIDITY IN LARGE	
	DEFLECTION ANALYSIS	
	B.4 MACRO4 FOR MODIFYING SECANT RIGIDITY IN LARGE	
	DEFLECTION ANALYSIS	
u	PENDIX C	
	C.1.1 A UNIFORM PORTAL FRAME SUBJECT TO A LATERAL FORCE	
	AND A VERTICAL FORCE 172	

C.1.2 A TWO BAY AND ONE STOREY FRAME SUBJECT TO TWO
VERTICAL FORCES AND ONE LATERAL FORCE
C.1.3 A UNIFORM FRAME SUBJECT THREE VERTICAL FORCES AND
ONE LATERAL FORCE
C.1.4 A UNIFORM BIG FRAME SUBJECT TO UNIFORMLY LATERAL
DISTRIBUTED FORCES AND VERTICAL CONCENTRATED FORCES 176
C.1.5 A NON-UNIFORM PORTAL FRAME SUBJECT TO A LATERAL
FORCE AND A VERITICAL FORCE
C.2.1 A UNIFORM PORTAL FRAME SUBJECT TO THREE VERTICAL
FORCES ON THE BEAM
C.2.2 A UNIFORM PORTAL FRAME SUBJECT TO A VERTICAL FORCE
AND A LATERAL FORCE
C.3.1 A CANTILEVER BEAM SUBJECT TO A CONCENTRATED FORCE AT
THE FREE END
C.3.2 A SIMPLY SUPPORTED BEAM WITH ONE ROLLER SUPPORT
SUBJECT TO A CONCENTRATED FORCE IN THE MIDDLE
C.3.3 A SIMPLY SUPPORTED BEAM WITHOUT ROLLER SUPPORT

List of Figures

Fig. 2.1 A Simply Supported Beam subject to Axial Forces
Fig. 2.2 Stress-strain Curve of Mild Steel [Neal, 1977]
Fig. 2.3 Elastic Perfectly- plastic Stress and Strain Relation
Fig. 2.4 Linear Relation between Load and Displacement
Fig. 2.5 Nonlinear Relation between Load and Displacement36
Fig. 2.6 Gloss Diagram (Sheshadri and Fernando, 1992)
Fig. 2.7 R-Node in Rectangular Cross-Section
Fig. 2.8 m-a method: Calculation of Reference Volume
Fig. 3.1 Beam Mechanism for a Portal Frame
Fig. 3.2 Sway Mechanism for a Portal Frame70
Fig. 3.3 Combined Failure Mechanism for a Portal Frame
Fig. 3.4Direct Secant Estimation of Plastic Limit Load [Adluri, 1999]71
Fig. 3.5.1 Portal Frame Subject to Concentrated Vertical Force
Fig. 3.5.2 Nonlinear Analysis Results for the Portal Frame in Fig. 3.5.1
Fig. 3.5.3 Direct Secant Method Results from Reanalysis for the Problem in Fig.
3.5.1
Fig. 3.6.1 Portal Frame Subject to Lateral Force
Fig. 3.6.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.6.174
Fig. 3.7.1 Portal Frame Subject to Both Lateral and Vertical Forces
Fig. 3.7.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.7.175
Fig. 3.8.1 Two-bay Single-storey Frame -A
Fig. 3.8.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.8.1

Fig. 3.9.1 Two-bay Single-storey Frame -B77
Fig. 3.9.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.9.177
Fig. 3.10.1 Two-bay Single-storey Frame -C
Fig. 3.10.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.10.178
Fig. 3.11.1 Two-bay Single-storey Frame -D
Fig. 3.11.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.11.179
Fig. 3.12.1 Two-bay Single-storey Frame -E80
Fig. 3.12.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.12.180
Fig. 3.13.1 Two-bay Single-storey Frame -F81
Fig. 3.13.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.13.181
Fig. 3.14.1 Two-bay Two-storey Frame82
Fig. 3.14.2 Nonlinear Analysis Results for the Problem in Fig. 3.14.1
Fig. 3.14.3 Direct Secant Results from Reanalysis for the Problem in Fig. 3.14.183
Fig. 3.15.1 Multistoried Frame Subject to Concentrated and Distributed Loads84
Fig. 3.15.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.15.184
Fig. 3.16.1 Non-Uniform Portal Frame Subject to Lateral and Vertical Forces for
Case 1
Fig. 3.16.2 Non-Uniform Portal Frame Subject to Lateral and Vertical Forces for
Case2
Fig. 3.16.3 Non-Uniform Portal Frame Subject to Lateral and Vertical Forces for
Case 3
Fig. 4.1 Segment of an Isolated Column
Fig. 4.2 Direct Secant Modification for Stiffness100
Fig. 4.3 Factor C for Portal Frames with Fixed Ends in Method 1101
Fig. 4.4 Comparison between Geometrical Nonlinear Analysis and Robust Secant

Method 1 for C=0.72	
Fig. 4.5 Factor β for Portal Frames with Fixed Ends in Method 2	
Fig. 4.6 Comparison between Geometrical Nonlinear Analysis and Robust Secant	
Method for β =0.22	
Fig. 4.7a Portal Frame with Concentrated Force on the Beam Case 1 (L2 < L1)105	
Fig. 4.7b Displacement Distribution in Reanalysis for the Portal Frame in Case1 105	
(negative for the left column and positive for the right column)105	
Fig. 4.8a Portal Frame with Concentrated Force on the Beam Case 2 (L2 = L1)106	
Fig. 4.8b Displacement Distribution in Reanalysis for the Portal Frame in Case2 107	
(negative for the left column and positive for the right column)	
Fig.4.9a Portal Frame with Concentrated Force on the Beam Case 3 (L2 > L1)108	
Fig. 4.9b Displacement Distribution in Reanalysis for the Portal Frame in Case3 109	
(negative for the left column and positive for the right column)Portal Frame Case 4	
The ALIG, Burgel Down of the Company of Press on the Barry Com A(12) (11) 110	
Fig. 4.10a Portal Prame with Concentrated Porce on the Beam Case 4 (L2 < L1) 110	
Fig. 4.10a Portal Plane with Concentrated Porce on the beam Case 4 (12 < 13)110 Fig. 4.10b Displacement Distribution in Reanalysis for Portal Frame in Case4111	
Fig. 4.10k Fortal Frame with Concentrated Force on the Beam Care 4(L2 × L1)110 Fig. 4.10b Displacement Distribution in Reanalysis for Portal Frame in Case4111 (negative for the left column and positive for the right column)	
Fig. 4.108 Displacement Distribution in Remarks for Portal France in Conservation (La Conservation Price 10) (Conservation (La Conservation Price Portal France in Conservation (La Conservation Price Portal France in Conservation) (Conservation (La Conservation)) (Conservation) (Conservat	
Fig. 4.100 Displacement Distribution in Remalysis for Portal Frame in Case4 (L2 < L1)	
Fig. 4.10: Topic-ment runne vana Categorianse releve on an other (La ⁻¹ L) - (La ⁻¹ L) (regardle for the left column and positive for the right column) for Vend Farme in Case4 — 111 (regardle for the left column and positive for the right column) Fig. 4.11: Proteil Frame with Two Forces on the Bram. Case5 (L2 < L1) — 112 Fig. 4.11b Displacement Distribution in Remarkyin for Portal Frame in Case5 112 (Lagneties for the left column) and positive for the right column) 113	
Fig. 4.10. Physicana Trans. Tests and a set of the s	
Trig 4. Not present Distribution in Remote Section 4. The Section 4. Let 2	
Fig. 4 Bry Len Displacement Conference in vector due were set. Let <i>s</i> -1, <i>s</i> -1	
Fig. 4.10: Diplement Tombus Antonio Festo da Paris Vice VI V	

(negative for the left column and positive for the right column)Chapter 5 Estimation
of Large Deflections in Beams
Fig. 5.1 Total Deflection of a Cantilever Beam
Fig. 5.2 Basic Secant Analysis
Fig. 5.3 Secant Analysis for Cantilever Beam129
Fig. 5.4 Nonlinear Behaviour of a Cantilever Beam
Fig. 5.5 Total Large Deflection of a Simply Supported Beam with One Roller131
Fig. 5.6 Behaviour of a Simply Supported Beam with One Roller
Fig. 5.7 Total Large Deflection of a Simply Supported Beam without Roller
Fig. 5.8 Simply Supported Beam without Roller
Fig.5.9 Comparison between Simply Supported Beams with and without a Roller134
Fig. 5.10 Portal Frame subject to a Lateral Force

List of Tables

Table 3.1 Re-analysis of Peak Moments Using Direct Secant Estimate for the
Different Structures Shown in Fig. 3.5.1 to 3.15.1
Table 3.2 Load Factors for the problem in the Fig. 3.16.1
Table 3.3 Load Factors for the problem in the Fig. 3.16.2
Table 3.4 Load Factors for the problem in the Fig. 3.16.3
Table 4.1 The Physical Data Used in Analyzing the Structures in Figs. 4.7 to 4.13

List of Symbols

А	a variation parameter
	area of member cross-section
ь	shorter side of cross-section
В	a variation parameter
С	factor used to predict elastic buckling load
$C_{lr} C_2$	arbitrary constants of the general solution of Euler equation
d	displacement
dat	arbitrary displacement
dape	displacement calculated from bending moment, total
	compressive force and factor C
dpeak	peak displacement
Ε	young's (elastic) modulus
E_t	secant modulus in i iteration
E_{θ}	young's modulus of original material
E_S	modified young's modulus
EI	flexural rigidity of beams and columns
Elald	flexural rigidity of beams and columns of
	original material
Elnew	modified flexural rigidity of beams and columns
$f(s_q), f(s_q^2)$	Von Mises yield function

F	externally applied force		
F_i	a set of forces		
$F(x_i^0, \sigma^0, m^0, \mu^0, \phi^0)$	function associated with Mara's function		
· ·	longer side of cross-section		
	moment of inertia		
-14	moment of inertia of the original cross-section		
	modified moment of inertia		
	iteration number		
	an arbitrary constant		
	citical value for octahedral shearing		
	stiffness matrix		
	length of beams and columns		
op	length of OD in Fig. 3.4		
a.	the horizontal length of beam in Fig. 3.1		
2	the vertical length of beam in Fig. 3.2		
n' m'	upper and lower bound multipliers		
e ni	upper bound multiplier corresponding to applied load		
,	structural property determined by Q		
n ₂	proposed lower bound multiplier		
1	bending moment		
d1	bending moment resulted from applied force P in Fig. 3.4		
d _{an}	bending moment at the location of zero displacement		
<i>d</i> _L	limit bending moment		
M _{dup} -peak	the bending moment at the location of peak displacement		

XVI

$M_{\rm max}$	the Maximum Bending Moment in Reanalysis
M_{\min}	the Minimum Bending Moment in Reanalysis
M_0	bending moment from an initial analysis
Monar	the maximum bending moment in the original analysis
M_P	plastic bending moment
M_{pi}	plastic bending moment for each element
Mpeak	peak bending moment
Mpeak-ave	peak average moment
п	an arbitrary constants
	number of peak bending moments
P	an element in a set
	applied load
Pcompesonal	total compressive force in the columns
Pilm	limit load
Q	structural property
	applied load
Q_e	effective generalized stress
Q_{L}	structural properties corresponding to limit load
q	modulus adjustment index
S	a set
SI	stress intensity
S_M	code allowable stress intensity
U	strain energy
V	volume of a component or a structure

shear force

V_R	references volume of a component or a structure
V_T	total volume of a component or a Structure
w	intensity of distributed load
W_E	work done by the external applied forces
W_i	a set of work
x, y, z	co-ordinates axes
Z	section modulus
μ	a constant proportionality
μ^{+}	plastic flow parameter
8	deflection
ε	elastic strain
e_{arb}	arbitrary strain in the modulus of elasticity softening process
\mathcal{E}_R	reference strain for reference stress
σ	elastic stress
σ^{n}	stress determined by the material and external load
σ_l	maximum principal stress
σ_2 and σ_3	principal stress
σ_{arb}	arbitrary stress in the modulus of elasticity softening
	process
(σ_c) mode	r-node equivalent stress
(a)m	maximum equivalent stress

σ_{ei}	Von Mises equivalent stress of the ith element	
σ_{a}^{z}	equivalent stress for any element	
$\sigma_{\rm s}$ and $\sigma_{\rm l}$	upper and lower stresses	
$\bar{\sigma}_{*}$	combined r node stress	
σ_R	reference stress	
σ_{κ}	R-node Peak Stress	
σ_{nf}	reference stress based on the theorem of nesting surface	
σ_{r}	Von Mises Stress	
$\sigma_{\rm so}~\sigma_{\rm y}$ and $\sigma_{\rm z}$	component of stress in a local coordinate system	
σ_y	yield stress (Equation 2-3 only)	
β	factor used to predict elastic buckling load	
$\gamma,\gamma_{1}\text{and}\gamma_{2}$	scaling factors	
Å	load factor	
2.	load factor for beam mechanism	
λ_r	load factor for sway mechanism	
θ	follow-up angle on the GLOSS diagram	
	angle of rotation	
δ_d	arbitrary displacement	
\mathbf{r}_{sy} , \mathbf{r}_{sy} and \mathbf{r}_{sy}	shearing stresses	
\$	curvature	
\$°	a point function defined in conjunction with yield criterion	

φ	degree of multi-axiality and follow-up
η	the location along the height of the cross-section
ΔV_k	volume of k th element
ΔM	the difference between $M_{\rm der}$ and M_0

Subscripts

arb	arbitrary
e	Von Mises Equivalence
y	tensorial indices
line	limit
у	yield

Acronyms and Abbreviations

APDL	Aansys Parametric Design Language
CSA	Canadian Standards Association
FEA	Finite Element Analysis
Gloss	Generalized Local Stress and Strain
R-NODE	Redistribution Node
UDL	Uniformly Distributed Load

Chapter 1

Introduction

1.1 General

Structure displaces consider as their pinnersy guid a safe and economical design. Note of the main insues in such a design process in the identification of all the potentia dimen mode of structure and the associated matume experision. Designers need to decise attractive and the associated matume experision. Designers need to decise attractive and the associated matume experision. Designers need to decise attractive and the associated matume experision. Design expected to use first and accounte methods use estimate the first load carrying capacities. Therefore, such methods are of single-first to incure correction.

Traditional approaches for the direct estimation of those maximum load capacities are developed on the basis of the two bounding theorems, viz., the upper bound theorem and the lower bound theorem as well as approximate step-by-step iterative formulation.

There are views techniques used for assessing "starth" limit load such as theoretical (cloud form)) methods based on minimizing one maximizing the upper or lower booknet respectively, the finite element motionic analysis and viscous other visburi methods. Theoretical analysis is easy to apply for simple structures but is not practical for complex structures. Even if their application is fourble, for most similarities the resolution are buffer. With the development of computer techniques, engineers are able to carry out complicated nonlinear analyses utilizing desktop computers in conjunction with finite element analysis (FEA). The FEA has been successfully applied in a great many fields. Finite element nonlinear analyses can be used to obtain the limit load for elaborate and quite complex structural problems. If the procedure is applied with ereat care and momenty verified, the results of nonlinear FEA can be considered to be accurate for practical purposes. However, it costs a great deal of manual effort in terms of care and verification as well as in computer resources (in spite of the cheap computing power available today, the discretization of practical problems is increasing in complexity steadily). To guarantee the accuracy, many iterations are needed. The control of convergence is sometimes difficult to handle. It requires engineers to possess a very strong background in nonlinear FEA and much practical experience to detect and avoid numerical difficulties. Even so, we need an independednt verification mechanism for the final results. Theoretical closed form solutions are obviously unavailable for complex structures to act as verifiers. In view of this, a resonably 'accurate', easy and fast method to solve the limit load problems is a major asset. 'Robust methods' try to fall in to this category.

Robut methods use once means finese analyses to solve multitum problems. Therefore, they are faste like-datic analysis and noi-ids the drawback that twellness multiply methods have, such as incremental handling problems, namerical instability and convergence difficulties. Using the vabout methods', researchers can hype to existant limit load copacities in a direct namera on solve problems that transformation depends on nonlinear analyses. The word "robust insplices that the multiplically depend on nonlinear analyses. The word "robust" implies that the analysis can withinked bigs overly seminor to nonmerical and dree difficulties valids at the same models of the solution of the models of the solution of the solution of the solution of the solution of the models of the solution of the solution of the solution of the solution of the models of the solution of the solution of the solution of the solution of the models of the solution of the solution of the solution of the solution of the models of the solution of the solution of the solution of the solution of the models of the solution of the soluti pominily speeds with an enteniev scope including different metural abares, boundary confidents and loading types. It would be advantageous to develop web robust methods for elementary design of components, planfic limit capacity colonizion, estimation of existed basis for budding, etc. It should be noted that for a paperso of the present anday. Timit lamit repletion the maximum balo capacity of the structure or components for a given set of properiors and load patterns. It is not the maximal limit tank tobuined by using load factors on the neurino load on the the maximum proceedings of the site of the structure of the structure and structure on the immediated one of the structure are designed for each in this state designed at REFD.

In Coasta, and structures are doing and apper CANCSAS-1691 [2007] and other the Canadas, The ARCL (2017) [2008] operitorized to used to 153. Also reads doing, Usually, structural limit state doings is performed using factored loads and their effects on individual members using classic analysis. This is questes the potential location which implies that there is no reserve arrangch the visibs reserved to an effect of the structure of the structure of the structure of coastion which implies that there is no reserve arrangch the visibs over the potential location which implies that there is no reserve arrangch the visibs are readed for structures. Allough iterative platmin analysis methods are repetitively they are array used in particular inclusion. Functional and which methods are reserved in analysis of the structure platmin analysis methods are reserved in a reserve and in particular inclusions. One of the main, formed in an using is form analysis of the structure platmin analysis methods are reserved in a structure. Allough iterative platmin analysis methods are reserved in a structure of the structure platmin analysis methods are reserved in a structure. Allough iterative platmin analysis methods are reserved in a structure of the structure platmin analysis methods are reserved in a structure of the structure platmin analysis methods are reserved in a structure of the structure platmin analysis methods.

Two main factors influence limit loads of steel structures: plasticity and buckling. Almost all the strength failure modes should include these effects. Therefore, it is necessary to focus on plasticity and buckling when assessing the strength of steel

3

structures.

The current study focuses on investigating the use of direct sectant modifications to member properties to estimate the limb shoul of finance structures subject to platicity and tability effects. An atompt will be made to probed large affection behaviour of learns, etc., sing these techniques as well. The method used is adapted from Athuit [1999] and Bolte & Athuit, [2006]. It is impired by existing techniques such as the games biage methods [Soid, 1977], the reade method [Sohabdi, 1997], and souldware TAA Thiles, 1986, etc.

1.2 Objectivities

- Adapt the direct secant method to framed structures and implement it in ANSYS software using APDL routines to carry out the estimation of limit loads due to plasticity.
- 2. Attempt to use the method to investigate critical loads due to elastic buckling.
- Investigate the use of the method to analyze the large deflection of simple beams.
- Compare the analyses of the direct secant procedures with those obtained from other methods and establish the applicability.

1.3 Organization of the Thesis

The organization of the thesis is briefly described bellow:

Chapter 1 gives general background, objectives the present study, etc.

Chapter 2 gives literature review. Some of the basic knowledge related to limit loads and stability in described. Next, traditional robust methods, such as R-Node method, m, method are reviewed. Lattly the principle of direct secant methods is generally discussed. Some of the relevant literature review is included in later chapters, as appropriate.

Chapter 3 focuses on the application of direct secant method to plantic limit load estimation. Review of plantic hinge methods in first given. The busic concepts such as plantic lingues and plantic colligner mechanisms are reviewed. The application of nobust direct secant methods on for plantic analysis in them introduced. The limit load of finand structure noused by plantic justice) in others introduced the con-

Chapter 4 describes the use of direct secant methods in solving buckling problems. Basic theory of elastic buckling is quickly reviewed. Portal frames with fixed supports subject to vertical and lateral forees are investigated. A proportionality factor is suggested to linearise the procedure. Critical loads for elastic buckling of different pond frames are solved.

Chapter 5 introduces application to large deflection analysis of simple beams. Direct securt method is used for approximately analyzing the large deflection of the beams (and sway of portal frames).

Chapter 6 quickly summarizes the thesis and outlines the main conclusions for the thesis.

Appendices give typical inputs for analysis using ANSYS software.

5

Chapter 2

Literature Review

2.1 Buckling and Structural Stability

Buckling in an inability phenomenon that results in adden fullisher without much warning. When relatively page members of structures are adjusted to axial propertience forces such as a large enough, the numbers will addenly suffer large lateral deflection leading to durative fullism. In practice, design codes derive formalise for columns with imperfections and slight initial load eccentricities that will be endered to large large large large labor.

Buckling load in referred to the solution resulted from imperfect columns which exist all over the world. The concept is used in actual cases, such as experiments. Another concept, called critical load, is known as the solution calculated from the prefect column. There is no "absolutely straight calsum" in the real world, so the concept of critical load is regulated in a theoretical value, and it is based on mathematical model.

2.1.1 Stable Equilibrium

If the elastic structure is applied a small enough external disturbances, it reacts simply by vibration about its original state, the equilibrium is stalt to be stable. In other words, although the small disturbances cause the vibration disturbancers. It is structure, the structure is able to maintain the original state after the vibration discoverse.

2.1.2 Unstable Equilibrium

If the elastic structure is not able to maintain the original state after it is applied the disturbance, the equilibrium is unstable. It will disturb the position of each point and tend to diverge from the changed equilibrium state.

2.1.3 Neutral Equilibrium

The boundary between the stable equilibrium and unstable equilibrium is called as neutral equilibrium. If the structure undergoes both stable equilibrium and unstable equilibrium, the external reason which makes this process is called as "critical", such as "critical load, critical moment and critical displacement".

2.2 Principle of Stationary Potential Energy

1 For any arbitrary displacement, the particle is in equilibrium if the total work done by all the forces acting on the particle is equal to zero.

Consider that a set of forces F_i act on a small particle, and it undergoes an arbitrary displacement δ_{dc} . During the displacement, each force acting on this particle will do a set of work W_{a} so the total work done by these forces is:

$$W = \sum_{i=1}^{n} W_i = (\sum_{i=1}^{n} F_i)\delta d = F_i\delta d + F_2\delta d + \cdots + F_i\delta d$$

(2-1)

If the particle is in equilibrium, the total forces acted on the particle are zero, which means that the total work done by these forces must be also zero. This results in the solution that the virtual work must be zero if the particle is in equilibrium.

2. For any arbitrary displacement, an elastic body is in equilibrium if the virtual

would date by the exittent fixers plus the visual work done by the internal fixers is equal to zero. The total visual work done by the fixers can be divided into two plus the visual work done by the exercise and which work done by the internal fixers. Board on the previous theory adors, if the datafic body is applichture, the total visual work must be zero. Therefore, the visual work work by the external fixers can be considered as group and correspondingly, the work done by the internal fixers can also be considered as another group. So it can be expressed as:

$$\delta W_{g} + \delta W_{f} = 0 \qquad (2.2)$$

3. For any and displacement, the class: structure is in equilibrium if so charge occur in the structures in equal in magnitude and expendits in the structures in equal in magnitude and expendits in the structures in equal in magnitude and expendits of the truth A(B) = -A(D). The total potential energy on this experiment is the structure is in equilibrium. The truth potential energy must be seens. If the attracture is in equilibrium matter of diagrams of freedoms, equilibrium must be enablished only usubficient of the diagrams of freedoms, equilibrium must be enablished only usual the truth potential energy does not charge for any possible drapers in the diagrams of freedoms, equilibrium must be enablished by requiring that no charge energy in the diagrams of freedoms, equilibrium must be enablished by requiring that no charge energy in the diagrams.

$$\delta(U+V) = \frac{d(U+V)}{dx} \delta x \qquad (2-3)$$

Since the dx is arbitrary, the increment of potential energy is equal to zero when

$$\frac{d(U+V)}{dx}\delta x = 0$$
(2-4)

2.3 Basic Approaches for Critical Load

Based one theory of equilibrium, critical land can be calculated in two ways. The first approach answers the question that at which had the neutral equilibrium is pushed. It is not seemany to dock whether the rimuture is studie or not. Intrad, one has only to enablish the equation to find the critical land and the studi potential energy is equal to zero. By requiring $d(t/t)^{-1}$, cf. critical land and the find. The second question is to determine that at which the lead at which the second question is used more than that of which the lead at which the second equilibrium to smallet equilibrium is pushed: It doub with the lead at which the second equilibrium, second a collaboration of the equilibrium and attradied equilibrium, second equilibrium can be determined by finding the land at which the second variation of the total potential energy changes from positive to sequive accoundialed by

$$\delta'(U+V)=0$$
. (2-5)

Ablough both of these approaches can be word to find the circuit law, they stay in discretated the first provide the start of the discretation between between the backling occurs, which means that the first approach pertains to "static level". Therefore, the first approach is usually referred to static approach. The second approach is static to static housday, referred to a state approach. The second approach is static to static the housday between the static housday maters which approach approach and the "sharing of the meanses, so the second approach pertains to "dynamic level". This approach is called as dynamic arrowch.

2.4 Upper Bound Theorem and Lower Bound Theorem

2.4.1 Upper bound Theorem

In mathemics, an upper band of a set 5 mean an element **P** which is proter or equal to every dement in the set 5. In structural analysis, the upper bound method is defined as: For eartimations which are adjusced to as of load fietor's every load factor *i* calculated from every possible failure mechanism must be greater than or equal to the load factor *i* calculated from the collipse mechanism. This concept is able called as "Knownet".

2.4.2 Lower Bound Theorem

An lower bound of a set 5, accordingly, means an element P which is less than or equal to every element in the set 5. In structural analysis, the lower bound method is defined are 1/m less data factor A is built and a matically admissible for certain frames, the value of load factor A is must be less than to equal to the load factor A corresponding to the collapse methods in Is it also called as Sinte Theorem;

2.5 Critical Load of Columns with Various Supports

2.5.1 Ideal Column

The issues involving structural stability are complex. The behaviour of an ideal column is well known and is reviewed below. The following assumptions are made for ideal column:

1. The column is perfectly straight.

2. Loads are applied along centroidal axis.

3. Material follows Hooke's Law and is homogeneous.

4. Column bends and bucks in a single plane.

5. Deformations of the columns are small enough.

2.5.2 Euler Equation

A hinge-hinged column is axially loaded, which is shown in figure 2.1. The internal moment at any locations with deflection y is

$$M = -EI \frac{d^2 y}{dx^2}$$

The externally applied moment is Py. Equating these two expressions:

$$EI\frac{d^2y}{dt^2} + Py = 0$$

This is homogeneous, linear, and second-order differential equation with constant coefficients. It can be solved using methods of differential equations. The general solution of equation 2-7 is:

$$y = C_1 e^{ib} + C_2 e^{-ib}$$
 (2-8)

Where $k = \frac{p}{FJ}$.

Using the relation :

$$e^{i\frac{2\pi}{3}} = \cos kx \pm i \sin kx$$
 (2-9)

The general solution is written in the form:

$$y = A \sin kx + B \cos kx$$
 (2.10)

In order to calculate the value of A and B, we need to give the boundary condition of hinged-hinged column:
$$x=0, y=0$$
 (2-11)

$$x=1$$
 $y=0$ (2-12)

After substitute these two equations, one can obtain the solution:

$$B = 0$$
 (2-13)

$$A \sin k l = 0$$
 (2-14)

A is not allowed to be zero because P can be any value under this condition. Since

 $P = \frac{\pi^2 EI}{l^2}$, P is the critical load needed to be obtained, $\sin kl$ is thereby required

to be zero. Then,

Substitute the expressions into the equation 2-10 leads to

$$=\frac{n^2 \pi^2 EI}{l^2}$$
(2-1)

P is obtained by setting n-1:

$$P = \frac{\pi^2 E I}{I^2}$$
(2-17)

This is the Euler load, It is the maximum load for the column capacity and it is on the verge of neutral neutral equilibrium. In other words, Euler load is the transition from stable to untable equilibrium. The column keeps straight surill external applied loads reach the Euler load. At Euler load, the column suddenly hows out and the lateral deflection becomes externed; large.

2.5.3 Critical Load of Columns with Various Supports

The table below depicts critical load of column with different boundary conditions.

Boundary Condition	Hinged-hinged	Fixed-fixed	Hinged-fixed	Fixed-free
Critical Load	$\frac{\pi^2 EI}{L^2}$	$\frac{\pi^2 EI}{(0.5L)^2}$	$\frac{\pi^2 EI}{(0.7L)^2}$	$\frac{\pi^{2}EI}{(2L)^{2}}$

2.6 Limit Load and Limit State

2.6.1 Limit Load

Limit load P_i is the maximum load which structures can take in service. It is equal to the product of load factor λ and external applied load P_i , $P_i = AP_i$. It is known that limit, load is proportional to load factor, which implies if the external applied load is constant, load factor, on he used to reactify the limit load.

2.6.2 Limit State

Basically, limits load gives the structures the limit state. Limit rate is fact in the mark where the facts laws have a fact limits laws. It more works were the facts are of the structures subjected to increasing load. Constrully, Limit rate despit includes two types: the fast is called aroungib limit rates and the durks in different limits of the data systems. The structure is a construct which be inclusion of 1 and employments are onlyno metadations during the structure of the data systems of the birth out the structure of the structure of the data system of the other limit, structure limit, and an attributive between of the birth out the other limit, structure limit, and the structure of the serviceability. The examples of the unacceptable serviceability involve deflection and corrosion.

The arranged lines state design is governed by the limit load datages which is carried out by the factor loads. When the factor loads reach the limit load, the structures are at allowed to have any pointial interright to stress the factor loads. If the structures are given the loads which are stronger than the limit loads, the menciones will colleapse. The outlinger mechanism because of plantic hinges can be succertailly solved by the scasses models. Based on the modified genomical properties, the reamalysis can predict the limit state and exclusion the limit loads. The results generated by scenar method are better than 95% or none. Instably including the circle calculation, because the motiopwork by the scenar method.

2.7 Elastic Perfectly- plastic Model

2.7.1 Stress-strain Relation of Mild Steel

It is necessary to isothodes the intersection strains of mild and The material is widely used in the construction of structures. The relation between stress and strain is mild stead in assuming its index stress stress and strains factor leads resuch the upper yield strength. This is shown in Fig. 22. The point arresponses the upper yield strength. This is shown in Fig. 22. The point arset wave stall makeday days down to the lower yield wave losses of the the investing external applied factors. The down is in the stress stress stress stress stress and arguing factor leads cause the investing stress losses of the strain which is in the region called area handwardy range. This is shown in region k. The maximum stress is warded at the societ, covered which areas (threas and then the stress downcers ward ruptures happen at d.

2.7.2 Elastic Perfectly- plastic Model

Engineers showed much instruct in the spid line on , which is indexible in Fig. 23. Sins seads in subards to also a more legible diagrams. The shope of the first choice is indexible of multiple states and the spin states and strain curves of multi state start the global point, because investments and a strain curves of multiple states and a global point. However, it is not state out the spin strain strain strain strain strain strain the strain strain strain strain strain strain strain strain the strain strai

The upper yield is not usually exhibited by some material, and the upper yield iters can not influence plantic moments. Therefore, the clanic performance plantic relation for stress-strain in often identified as the sequent applied lands can not make the stress increase anymore. The stresses keep constant while the strains keep increasing until immibility courses. This is stressed the slated plantic relation or elarnic perford plantic model.

2.8 Failure Criterion

2.8.1 Maximum Principle Stress Criterion

When the maximum principal stress reaches the uniaxial yield strength, yieldding will occur.

$$\sigma_1 \le \sigma_2$$
 (2-18)

(2.21)

2.8.2 Von Mises Yield Criterion

The Von Mises stress is expressed as:

$$\sigma_{\gamma} = \sqrt{\frac{(\sigma_1 - \sigma_1)^2 + (\sigma_2 - \sigma_3)^2 + (\sigma_1 - \sigma_3)^2}{2}}$$
(2-19)

in which σ_i , σ_j and σ_s are the principal stress in three directions. In one dimension case, this stress becomes uniaxial stress. If the expression is based on a local coordinate system, it is written as:

$$\sigma_{z} = \frac{1}{\sqrt{2}} \sqrt{(\sigma_{z} - \sigma_{z})^{2} + (\sigma_{z} - \sigma_{z})^{2} + (\sigma_{y} - \sigma_{z})^{2} + 6(\tau_{w}^{2} + \tau_{yz}^{2} + \tau_{w}^{2})}$$
(2-20)

Plastic yield occurs when the von-mises stress or equivalent stress reaches the yield strength. Von-mises stress can be used to predict the failure of ductile tearing.

The vor-misses yield criterion is based on the concept of maximum distortion strain energy. It states that failure occurs when the energy of distortion reaches the same energy for yield or failure in uniaxial tension. Mathematically, it is expressed as:

$$\frac{1}{2} \left[(\sigma_1 - \sigma_2)^2 + (\sigma_2 - \sigma_3)^2 + (\sigma_1 - \sigma_3)^2 \right] \le \sigma_y^2$$

In the two dimensional situation, one of the principal stresses is zero: $\sigma_j = 0$ and the Von-mises yield criterion reduces to the extremsion:

$$\sigma_1^2 - \sigma_1 \sigma_2 + \sigma^2 \le \sigma_2^2 \qquad (2-22)$$

We can also interpret the von-mises yield criterion in term of octahedral shearing stress. When octahedral shearing stress reaches the criterion which is defined by the critical value, the material vields.

$$(\sigma_1 - \sigma_2)^2 + (\sigma_2 - \sigma_3)^2 + (\sigma_1 - \sigma_3)^2 = 6k^2$$

(2-23)

Where k is the critical value and it means the yield stress in the case of pure shear. If the loading is uniaxial, the above equation can be reduced to:

$$k = \frac{f_s}{3}$$
(2-24)

2.9 Nonlinear Analysis of Structures

In the finite element technique, the relation between load and displacement is given by:

$$F = Kd$$
 (2-25)

is shick F is the extensity applied land matrix, K is the stiftness matrix and d is the displacement matrix. Whether the analysis of attractures is linear or solutioner depends on the stiftness matrix. If the origination of the storest derived the entire analysis in a called linear analysis (Fig. 2-6), otherwise it is notificare (Fig. 2.5). Linear analysis implies stiffness matrix is not changing. Nonlinear analysis in contrast, tanda to changing atfittees matrix. In other works, the fundament is contrast, tanda to changing atfittees matrix.

It is known that stiffness is related to Young's modulus, moment of inertia and the

length of modelses. Any changing of them occurs among them loads to motivate behavior of structures. The changing of Yoong's modulus length to material molitariation. Genomics and/material results from the changing of genomic properties, such as moment of horits and length of members. Another lixed of motification is caused by the changing strans which licklade the changing of Souly, coasta form, or offer command members.

2.10 Limit Load Prediction

Traditional methods of analyzing the limit load utilize the upper and lower bound theorem. If high degree indeterminacy structures are involved, the methods are much more time communing and tedious and they are not always practicable for complicated structures. Therefore, quick and accurate methods are needed.

Over the year, IFA has been successfully applied to structure analyses, it is very peptihe because of a universality and generality. There are analyses, it is very peptihe because of a universality and generality. There are and the changes of structures if project elements and techniques are chosen. The analyses can be taskic or dynamic. However, if the structures below inclusionally, linear onlyses on network and mellinear analyses, there have the structure below inclusionally, linear onlyses on network and and mellinear analyses are needed. TEA is able to deal with the nonlinear problems but it has drawbacks. First, because FIA was itsneive data calculations to do the confinem multyses are at momest of compare resources and time are needed. Second, convergence methins are defined for the outpare mechanism, so a structure difficulties may appear. Mang factors can influence the sensition, such as one types or dimension, Laidt contegriting from the first. Laidt change from the sensition. Laidt changing from the sensition of the sensition of the sensition. Laidt changing from the sensition of the sensition. Laidt changing from the sensition of the sensition. The sensition of the

above factors may produce significant differences in the solution. Therefore, it needs analysts to have much work experience and expertise. Researchers are thereby encouraged to develop simpler methods and simplified software for practical application.

The techniques used for semitors analyses generally can be divided into two categories: they are tangent and secant stiffness method. One typical tangent stiffness method is Newton-Raphons technique. Nevers methods include the direct secant method and the incremental secant method. Robust methods are one of the direct secant methods. Various Robust methods are developed to analyze the instantic effect based on netwice baseds.

The nobstanchols are gowerful stabilized of analyzing the finite bodt. They are none popular and arrays than finite disconst tablement analysis when work for complex structures. This chapter describes a set of efficient techniques used for analysing the limit had. Instead of using finite denoses to molinear analysis, there exhibits the averall measures to prictic the time of the model on a very small number for industry to many the time of the time of a model on a very small number for industry to prictic the time of numerican barreness. Bounce frow the time and the set of the set of the time of the set of the set of the Second, southings analysis structures have none time than traditional settochs. Second, southings and using a dark the set of the the single of the direct or detaining a solution, and many mathematical techniques are utilized to define are writtens to make uses finite denome smallpare margins on weak in complexes. give complexited convergence entries and can obtain a final toution with impact exclusions. Thus, another important significance of robust methods but can be used as a entries to collarity the train of any nonlinear analysis. Decause they are relatively upids and accurate, now can judge the result generated from the entries of the training of the entries of the entries of the can net ANSY 56 which entries the same methods. For example, people can use ANSYS is the nonlinear analyses, but to theoretical oblicitoms or gives to judge the result of ANSYS. Robust methods give goed theoretical estimation, so the members can tacked the results to sing them.

2.10.1 Reference Stress Method

It was recognized that in a energing hom, interest a certain disklead point. Exp constant. Soldware, (1441) in his experiment first from the some retreews key constant in a softeness space. This constant for multi-solid cerup deformation and observed that even though the system is widely softened and stresses are redistributed in a larger range due to schedule, there are still some schedul stresses which always memiated constant. Lates, the reference reress for promate venech was obtained by him. Schedul (1960) observed that in a coreging hom, there were two points in the addression of their points. Marcint (1961) and Lackke (1964) observed that are stresses of exist shale shaden goes be transient eresp target constant with finar. Three points were defined a schedur points. Sime (1974) immeddeed the analylical defining for furthermeter stress. Since for reference stress, in independent of the energy Sime advance of a stress defined provides. Sime (1974) immeddeed the analylical defining for furthermeter stress. Since for reference stress, in independent of the energy Sime advance of a stress defined provides that there advances that is independent of the energy. certain cross-sections. Therefore, the stresses of infinity exponent croep are analogous to the yield stress, and thus the reference stress is expressed as:

$$\sigma_{g} = \left(\frac{P}{P}\right)\sigma_{g}$$

(2-26)

where P is the applied load, P_i is the limit load and σ_i is the yield stress.

2.10.2 Partial Elastic Modulus Modification

Matisti (1984) developed a technique used to identify the interms individually generated from poor yield areas. This method includes performing a sequence of initial markows and the two-board detorms. The inverses caused by the factured loads are genere than the yield stress, the iters in redshifts of the two-board detorms. The methods of each denser should be modified according to the quantiset. The methods of each denser should be modified according to the quantiset. $E_{i} = E_{i} \frac{S_{i}}{M_{i}}$, where S is the maximum equivalent stress calculated from the previous interior, S_{ij} is the arbitrary stress, and S_{ij} is the code aboved to ress.

Securit intensions are then run based on this modification. Since it should be been than the code allowable stress, then maximum equivalent stress, after securit intension. The concentration of the stress stress is the stress of the stress of the stress advantages of the stress advantage of them the code advantage value. The equivalent stress calculated from the previous calculation should be statistically advantages, to the fastered load thus can be considered as a lower board of the limit load. This reduced modules method is aliand to find the maximum equivalent traves for factored loads rather than find the limit hand. Thus the reduced modules can not somer that the coverage three is always loss than the allowable struss. In should be noted that only specifical pertisms of structures undergo the modified moduli, therefore this module does not totally describe the areas reduktivation during plantic collepter.

Further work has been extended by Seehaldi (1991) and his co-workers (Fernando,1992; Mangalaramana,1997). The R-loade method has been developed by them to predict the limit load. R-Node stress and repeated elastic moduli modification are used in the R-loade method.

2.10.3 Gloss R-Node Method

Based on theory above, the Gian K-Mold Meldod was introduced by Selabaki in 1991. "Glane" is an account for "Generalized Local Stress Strain" and "Noide" howas as the "Reliabated Noide". This method is used to evolvate the approximate limit hand for both plattic soutinearity and material southnessity. It is a substat and efficience schedups based on two datafic finite element analysis which common deministrate R-Noides. Reliabates are introduced and the latest interface and the structures, and some of the R-Noides peaks, acting as the possible plattic hange locations, finally predict the plattic collapse mechanism and the limit housis two obtained.

2.10.4 Gloss Method

There are two types of controlled network land controlled areas and deformation controlled network. Land controlled networks result from static deformation articulter are analysis the structures to karp the network result of the networkers are subject to external applied forces and moment. Deformation controll draws, solverse, coccurs in the articulers in the number of the structures and solverses, the structures and the number of the structures result. One the structures undergo the platicity or earces, the stratically industruminate articuments that have bayes at most position of the structures course it entitle location, which are known as R-Node locations. The structures under conditionation can be divided into two signatics, local regions and mainford region. The local regions of the structure analogue inclusive deformation, such as platicity and curys. The remainfore results that the domainstice, such a platicity and curys. The remainforresults that the domainstice deformation are such as the structure analogue in the structure analogue in the domain of the structure analogue in the domain of the domain of the structure analogue in the domain.

The principle of the Gloss Method is to utilize during analyses to colorate interintity. The inductor stress millimbation due to planticity or encore on the onlyzed by dowing mixed areas relation. In one during predencity plantic model, stresses relate to yield stress due to planticity. If deformation control govern, it will keep the strain as contrast value which can be determined by requiring # to be equal to zero. To source the effect of help lanticity and deformation control, the modulus of this produc durindly stress denses in solified by the experision:

$$E_g = E_g \frac{\sigma_{ab}}{\sigma_a}$$
(2-27)

 E_i is the new modulus for each element, E_i is the original Young's modulus, σ_{ee} is an arbitrary stress value and σ_e is the equivalent stress for each element. After modifying the modulus, a second analysis is then run. It was suggested that this reduced modulus method predict the limit load with enough accuracy.

Figure 2.6 presents the Glaum diagram. Line OAC is the claimic perfectly plantic curve for stress and strain. Line OAB is the preseds educit line on which the first ducit malysis is hands. First B (σ_{max}, c_{max}) is the preseds point. Deformation Control is performed from the point B. The slope of line OCE is the new slope which is modified from the original slope of line OAB. The second educit analysis is performed by using the new slope of line OCE, which is called the secont modulus.

2.10.5 R-Node Method

Schalds in 1911 introduced an approximate method and to determine the limit liable based on two durine, and yess. These method, have one is Guine R-Nob for method, is inspired by the reference attess method and modulus modification. When structures mecoastare inspiration, such as planticity and array, structure distribution will court in the periodic of the structures undergo stress methodichonics on electron at a Node location. K-Nodes always maintain the same rener level fluor institution of the structure of the structure structure in the strucmentitution of the structure of the structure structure. These locations are known at a Node location. K-Nodes always maintain the same rener level fluor structure indications are structure or distributions while no distribution impersent and the structure of the structure structure indirect of matterials. If they different content loads are applied on metatrase and triping the relatively. If the different content loads are applied on metatrase and triping the relatively of the structure of metatrability indirect at the Node and triping the relatively. If they different content loads are applied on metatrase and triping the relatively of the structure of the Node and the specification is the structure of the Node and the specification is the structure of the Node and the specification is the structure of the Node and the specification is the structure of the Node and the structure of the Node and the structure of the Node Andre Andresson and the structure of the Node and the structure of the Node Andresson and the structure of the Node Andresson and the structure of the Node Andresson and the Node and the Node and the Node and the Node Andresson and to locate R-Nodes [Seshadri, R., 1991].

Imagine that a beam with rectangular cross section is subject to pure bending. It is shown in Fig. 2.7. The relation between stress and strain is expressed as :

$$\varepsilon = E\sigma^*$$
 (2.28)

in which E and n are determined by the material and external loads. If the cross section behaves clatically, n-1 and if the cross section is totally plantic, n=n. The intersection of these two lines is the location of the R-Node and It is recognized that all the stress redistributions pass through the same nodes (Mangalaramann and Schahl, 1997).

This method suggests that except at R-Node locations, all the stresses redistribute due to plasticity within components or structures. In the elastic perfectly plastic model, the relationship between the reference stress and the R-Node stress is given by:

$$(\sigma_{e})_{e-mak} = \mu \sigma_{g} \qquad (2.29)$$

Since the induced stresses are proportional to the factored loads or load combinations, this relationship can be given by :

$$(\sigma_c)_{r-mak} = \gamma_1 P \qquad (2.30)$$

$$(\sigma_r)_{r-mak} = \gamma_2 < P, M >$$
 (2.31)

where γ is the scaling factor determined by loading, material, and geometrical properties. In an elastic perfectly plastic model, when the induced stress reaches the yield stress, the factored loads will become the limit loads. Therefore, this relationship can be expressed as:

 $(\sigma)_{c} = \gamma_{1}P_{L} \qquad (2.32)$

$$(\sigma)_{y} = \gamma_{z} < P_{L}, M_{z} >$$
 (2.33)

Combine equation (2-13) and (2-15) :

$$P_{L} = \left[\frac{\sigma_{z}}{(\sigma_{z})_{z=ab}}\right]P$$

$$(2.34)$$
 $< P_{z}M > \left[\frac{\sigma_{z}}{(\sigma_{z})_{z=a}}\right] < P_{z}M >$

Where
$$\left[\frac{\sigma_y}{(\sigma_z)_{z \to adit}}\right]$$
 is identified as the load factor.

The R-Node method can be used to analyze the limit load of mechanical components and structures in the following steps:

 A linear clastic analysis is performed by the factored loads, which can be greater than or less than the actual limit loads of the structures. This analysis is the pure clastic analysis without any limitation for the structures, such as yield stress and buckling. Sublify is not under consideration for the analysis.

The modulus of each element is modified by the equation: *t_i* = *t_i* = *σ_i*, *σ_a*, *σ_a*, *σ_a*, *σ_a*, as mentioned before, is an arbitrary nonzero value. According to this modification, the second elastic analysis in then carried out.

 Two efficies analysiss are performed and they result is two static lines which are an the basis of the follow-orp angle θ. The locations with θ = 90⁻¹ are identified an the Node bocstoom. Antually, Node means are resultabilistin occur for the stress, so if the intersections of these two analyses can be obtained, R-Node location can antatully be found. The stresses at the R-Node locations with stress.

- 4. A plot of R-Node peak stress identifies certain locations within the structures. Theses beatines imply that as the externally applied load increases, the cross section of the peak tress locations will become totally platic faster than ambient errors sections. In other words, the peak stress locations form platic hinger faster than the neighbor points.
- R-Node stresses are the results of load control. The limit load thus can be calculated when R-Nodes stress reach the yield stress.

$$\overline{\sigma_s} = \frac{\sum_{i=1}^{n} \sigma_{si}}{N}$$

(2-36)

Where σ_z is the yield stress and $\overline{\sigma_z}$ is the peak average stress.

Compared with other instantic methods, R-Node method gives relatively simple procedures and conservative results. It is successfully applied on two dimensional situations, For three dimensional structures, it is suggested that a R-Node stress surface should be used to determine the near R-Node locations (Senhaft R, 1997).

2.10.6 The m-a Method

An improved limit laad entimate tuebringe impired from Murri's variational formalisein is known as the m-a nethoda (Sohahai and Mangaharamun 1977), also of the obstation of heavy variational formation, the limit laad is indived by leapfrogging on a huis of two limour charies analyses which result in the upper and lower bound multipliers: a⁰ and m². Similar to the R-block method, the m-a method modifies the initial dentic moduli for each closent in order that strenges can multiplicate and and anomality laboration. an upper bound multiplier m² which satisfies the theorem of neuring surface (presented in 21.057) is determined based on the instance actions of the structures. Compared with the m₁-obtained mone he total volume of structures, the one estimated based on this new upper bound theorem is more constructive and advanced. The main procedure used for testinging m₁, is contained as the following procedure

- The first elastic analysis is performed to analyze the stress distribution which is utilized for modulus modification.
- 2. According to the modification equation

$$E_{\mu} = \left[E_{\mu}\frac{\sigma_{\mu\nu}}{\sigma_{\mu}}\right]^{\mu}$$
(1)

all the element moduli of the structures are modified, in which E_0 is the Young's modulus, σ_{os} is an arbitrary stress , σ_a is the equivalent stress and q is the modulus adjustment (usually chosen as 1)

- The second linear elastic analysis is carried out based on the new modified moduli and a new equivalent stress distribution is evaluated.
- 4. Calculate the energy dissipation of each element in the prescribed structures and estimate the upper board as⁴ for each element. Two conservative analyses give two multipliers doubted as m²₁ and m²₁. Plotting there two curves gives the intersection which is identified as the reference volume where the theorem of next structure is stuffed [92, 23]. The upper yourd multipliers it thus obtained.
- 5. The lower bound is given by the equation:

$$m^{*} = \frac{2m^{0}\sigma_{y}^{2}}{\sigma_{y}^{2} + (m^{0})^{2}(\sigma_{z}^{0})_{W}^{2}} m_{y}$$

2.28)

 $m'm^0$ and $(\sigma_1^0)_{ij}$ are all functions of the iteration variable. Based on iterative

calculations, the multiplier m_p in the end can result in good estimation of the reference volume.

2.10.7 Nesting Surface Theorem

Calladine and Drucker in 1962 introduced the "theorem of neuring surface", which is sured to estimate power law corep. Boyle in 1962 redefined the theorem which is used to simplify the analysis of atress in complex structures. The sverage energy dissipation rate is utilized as the expression of the dissipation rate of structures under multiple looding.

$$\sigma_{\pm}\varepsilon_{\pm}V = \int_{V} \sigma_{\pm}\varepsilon_{\pm}dV$$
 (2.39)

The material of structures is given by the equation:

$$\epsilon = E\sigma^{*}$$
 (2.40)

Using equivalent stress and strain, the average energy dissipation can be expressed by:

$$\sigma_g^{\mu\nu} V = \int_V \sigma_e^{\mu\nu} dV \qquad (2.41)$$

Therefore, the reference stress can be written as :

$$\sigma_R = Q_e(\sigma_g) = F_e(\sigma_g) = \left[\frac{1}{V}\int_V \sigma_e^{sel}dV\right]^{\frac{1}{sel}}$$
(2.42)

The function is articity monotonic with the component n. When n-1, the structures behave calcularly and the function is at lower boundary. When $n \rightarrow \infty$, the function is perfectly platic and the function is bounded above. At this time, Q is consistent as a constant value. Thus if an arbitrary stress having a hypersurface Q--constant, must "star" inside the regime between the opera only lower boundary. The reference stress defines stress space with two boundaries or two surfaces. When n=1, it is bounded outside and when $n \rightarrow \infty$, it is bounded inside.

$$Q_{c}|_{m} \le Q_{c} \le \lim Q_{c}$$
 (2.43)

2.10.8 Extended Lower Bounded Theorem

Mura and Lee in 1965 introduced a lower bound theorem used for estimating limit loads of structures subject to tensile loads. This theorem is based on variational principle and gives good evaluation of limit load. However, real structures are more complicated because they are not only under tension. So a generic approach is necessary. Seshadri and Manealaramanan in 1997 proposed a method which combined the elastic modification and lower bound theorem. The evaluation of limit load is on the basis of elastic stress distribution.

Mura and Lee demonstrated that if the function

$$F = m^2 - \int_0^1 \mu^2 [f(s_s^2) + (\phi^2)^2] dV$$
 (2.44)
 $f(s_s^2) - \frac{1}{2} s_s^2 s_s^2 + \frac{1}{3} \sigma_s^2$ (2.45)

where

is stationary, the factors m^0 , μ^0 and ϕ^0 can be determined. This leads directly to the following three equations:

$$\frac{\partial F}{\partial \mu^2} = 0 \quad \frac{\partial F}{\partial \mu^2} = 0 \quad \frac{\partial F}{\partial \phi^0} = 0$$
(2.46)

Evaluating leads to

(2-47)

(2-45)

$$m^{0} = \frac{\sigma_{y}\sqrt{V}}{\sqrt{\sum\limits_{k=1}^{n} (\sigma_{ak}^{k})^{2} \Delta V_{k}}}$$

Comparing the expression for m⁸ with the one proposed by Calladine and Drocker in 1961 and Boyle in 1982 which is obtained from the reference stress equations, it is given that:

$$m^{0} = \frac{\sigma_{\gamma}}{\sigma_{z}}$$
(2.49)

(2-48)

Therefore, monotonic increasing reference stress will lead to the decrease in the value of m^2 . Since it shows a lower bound of the reference stress when $n \rightarrow \infty$, the m^2 is thus a super bound multiplier when m-1.

Mura in 1965 proposed the lower bound theorem:

$$n^{2} = \frac{m^{2}}{1 + \max \left\{ f(s_{e}^{2}) + (\phi^{2})^{2} \right\} / 2k^{2}} \le m$$
 (2.50)

This equation can be simplified by substituting equation (2-45) as:

$$m^{2} = \frac{2m^{2}\sigma_{j}^{2}}{\sigma_{j}^{2} + (m^{2})^{2}(\sigma_{s}^{2})_{M}^{2}} \le m$$

(2.51)

in which $(\sigma_r^0)_{\mu}$ is the maximum equivalent stress for certain factored load. The limit load can thus be estimated by the equation:

$$P_{im} = m^* P$$
 (2-52)

According to the upper and lower bound theorem and the solution obtained above, the limit load bound is evaluated by :

$$m^{2} \le m \le m_{2}^{0}$$
 (2-53)



Fig. 2.1 A Simply Supported Beam subject to Axial Forces















Fig. 2.7 R-Node in Rectangular Cross-Section

(Sheshadri and Fernando, 1992)



Fig.2.8 m-a method: Calculation of Reference Volume

[Seshadri and Mangalaramanan, 1997]

Chapter 3

Plastic Limit Load Estimation

3.1 Introduction

Structures subject to relatively small loads behave elastically. When larger loads are applied, they exhibit plastic behavior. In frames and frame like structures, plastic zones and eventually plastic hinges are formed with increasing loads. This leads to the redistribution of stresses. A plastic collapse mechanism takes place if enough plastic hinges are formed. This procedure can be analyzed by traditional plastic analysis techniques. Neal [1977] developed a method using minimization of work done to determine the final plastic collarse mechanism. This method can give good estimation of limit loads for simple structures. For more complicated structures, it becomes tedious. One has to determine the collapse mechanism by comparing all the independent and combined mechanisms especially when the plastic hinter locations are not obvious at the outset, e.e., frames with unevenly distributed loads. The method also completely ignores the effect of lateral displacements. The alternative is to carry out full scale nonlinear finite element analyses which require complete discretization of the entire cross-section throughout the frame. The mesh needs to be especially fine near the locations of plastic hinges. Although the computational time needed for very fine meshes is no longer an overriding issue. complete discretization of the frames needed to capture the progressive plastification of the cross-sections at several locations in the frame is tedious work requiring initial input and extensive cross-checking. Simplified techniques therefore become attractive. Such methods should be progressively more advanced and yet be simple, accente and robut. The rende method [Sechtaki, 1997]; steadaki, 1997] 1992] is one such technique as settimed in the previous chapter. The Direct Securit Medical is a new technique which is impired by both the traditional planic analysis and the rende method [Aulat, 1999a]. In linn method, based on the solutions of the first chartic analysis, nightfies are modified and are used to oray out a second analysis. The rendes the second chartic analysis can be utilized to predict the limit both of structures.

This despite fact illustrates the principles of radiotical pionic analysis. The concepts of planic monota and collapse mechanisms are reviewed. Three bain collapse mechanism, vize, beam mechanism, your mechanism and combined mechanism are explained. After listing the impérations of tradicional planic method (will be presented in 3.6) and the N-bole method, this despite explains the main steps of the direct sector method. In these explositions counties are about, a

3.2 Plastic Analysis Review

The matrix insufance analyses of structures have been investigated by recursive for many years. It is well known that the redistribution of structures council by placing can generate the strength which can be end to a being the structures byoard the single 'data's limit, "Katistory [1914] was one of the early researchers in speec experiments on investigating the outputs of a stort bown. He found that the collespon of this fixed back back account which cannow sections are the middle and the end became fully plantic. The moment which cannot the environment of the became fully plantic is called the plante moment. When a sense-storin yield highly, in on stort like a bank. Noveal thatous in structures are not dive to take any

41

moments. However, the hinge caused by a plastic moment, known as the plastic biney can resist the plastic moment but would not have any further moment resistance. The formation of plastic hinges can lead to the collarse of structures. As loads increase, enough plastic hinges form leading to the collapse of the structure under consideration. Just prior to the collapse, the state of the structure, including the locations of elastic binnes, etc. is called the collarse mechanism. The collarse mechanism is assumed to be the final state of the structures [Neal, 1977]. The loads leading to the collapse mechanism are termed as limit loads for the purpose of the current study. The limit loads depend upon the load configuration as well as the structure's properties, geometry, boundary conditions, etc. Thus, for a given structure, there can be several limit loads depending upon the type of loading patterns. For a given pattern, there is a unique limit load. The limit load is usually identified by a multiplying factor applied to increase the nominal load to the limit load. This factor is called as the load factor in the current study. The determination of the load factor and the associated failure mechanism is the purpose of limit analysis and design.

Structural design, nost often, ignores the refinit/biolist efficient that occur byout the classel limit (excerpt some cases, ands a axianic design). The results of classic analyses are used to design the momber enservations in a platitic some. This generally implied that as soon as the most critical location reaches its limit, the structure is unsuble over through the most critical location reaches its limit, the structure is unsuble over through the most critical location can support refortibulion that againfancting improving the limit load. This critica capacity unsubje cover the structures that available indemntion. The process of the platics design in thereby

42

to find the loads which cause the finiture by successive formation of plancic langes in a mechanism failure pattern. For the current study, it is assumed as per normal convention, that the effects of combined moment and shear are not significant. It should be noted however, that this is not a necessary condition for the method being studed. Even if them defermation dominate, the method can be adapted.

The plastic moment capacity, M_{p} , of any given cross-section can be calculated using geometrical properties and the yield stress. For the rectangular cross-section, the plastic moment capacity is,

$$M_{\mu} = Z\sigma_{\mu} = \frac{1}{4}bd^{2}\sigma_{\mu}$$
(3-1)

where, Z is the plastic section modulus, b is the width and d is the height of the cross-section.

In general, the plastic modulus can be obtained using

$$Z = \int b(\eta) d\eta \qquad (3-2)$$

where, the cross-section which can change with the location *x* pairing the height. The traditional method to determine the limit hand of finnes had been to use the appentored downers, quaring the work done by the late to the work absorbed in the plattic larges for a section of possible mechanism and selecting the least limit had factor. If the mechanisms depend on plattic hange locations that are continuously changeable, a simple minimization process is carried out. This is illustrated in securit andated references such as NeW [1977].

3.2.1 Mechanisms of Failure for Frames

For frames, three possible mechanism types will be shown in this section. Fig. 3, 1 shows a beam mechanism where a portal frame is subject to a vertical force in the middle of the beam and the failure of the frame is essentially through the failure of the beam as shown. The end hinnes can be located either in the beam itself or at the ends of the column. As the load increases from a completely elastic state of stress, stress at the extreme fibers of the middle of the beam first reaches yield. An increasing load then makes the engancection in the middle become totally plastic and turns it into a plastic binge. A redistribution of any further increase will prompt the end of the beam to start yielding and eventually to the formation of two additional hinges. Since three hinges in a line imply instability, the structures will then reach its limit load. Equating the work done by the arefied factored load (P 2) for a given rotation (A) of the beam seement, with the internal work done due to the rotation of the cross-sections at the plastic binnes (thus isnoring the small amount contributed by the elastic deformation of the remaining structure), we can obtain the load factor λ_4 for the beam mechanism as

$$(\lambda_{\nu}P)\left(\frac{I_{1}}{2}\theta\right) = 4(M_{\nu})(\theta)$$
 (3-3)

The deflection in this case is assumed to be small so that the displacement under the load is directly calculated using the rotation of the beam.

$$\lambda_5 = \frac{8M_p}{PL}$$

Following a similar procedure, the sway mechanism is shown in Fig. 3.2. Imagine that

a portal frame subject to a lateral force which can cause the lateral deflection. The collapse mechanism include + hinges for instability. If the cross-section capacities are all equal to M₂, equating the externally applied work and internally absorbed work gives,

$$(\lambda_r F)(L_2\theta) = 4(M_p)(\theta) \implies \lambda_r = \frac{4M_p}{FL_r}$$

(3-5)

Both beam and away mechanism in the above are assumed to be independent which do not influence each other. The limit load factor (λ) generated from these mechanisms follows the upper bound theorem.

$$\lambda \le \min(\lambda, \lambda)$$
 (3-6)

The exceptes given are simple pound fusion. In a real plantic analysis, most increases encountered have complicated geometrics. Thus the combination of those independent mechanisms in a way to solve plantic problems. As an illustration (Fig. 33), if dis pertuil makes in analyzet to both the latent and vertical force, we can see that most rotations in the combination of the two mechanisms exist, while two furgepear. This is because the rotation sign at these locations in the twom mechanism is opposite to the ourse and two modersing in the rotations. The rot strice.

$$\lambda P\left(\frac{L_1}{2}\theta\right) + \lambda F(L_2\theta) = 4(\theta)(M_r) + (2\theta)(M_r)$$

(3.7)

The external work is equal to the sum of that in the beam mechanism plan that in the rawy mechanism. This implies that the patterns of work done by the applied loads are independent and do not influence each other. On other hand, the internal maximum work shorted in not atmice cault to the sum of the work done by the two the start of the independent mechanisms. Therefore, the load factor calculated based on plastic hinge cancellation may be smaller than that for each independent mechanism, [Neal, 1977].

The method outfined above gives and estimates of the limit hand for fitteen type structures. However, the protection is validate of the method requires us to locate all the large ensemblation and calculate the exact maximum instantal work, which an ore usy handless especially if the fitner has had, distributed in a maxement anazer. The method is also are analy anomable for computer implementations. The method all some analyments the fit of columns may on the column moments that to verifical hands (instruction between the tway forces and the column moments that the verifical hands (instruction), bolds, is some cases may retributed from longing to mooned amplification, bolds, is not cases may retribute the long factor to a catting exact, the needed gives excitent results in most cases and its work for for compution protects.

3.2.2 Plastic Design of Structures

Planic analysis in of significance for dospings not and other kinds of nurearcess that have redundancies and can withstand exits deformation beyond the initial sided. If the collegen mechanism of structures shows not include a sway mechanism or does not include local and overall stability effects, the planic analysis is called "first order planic analysis", such as in the case of a flued-fixed beam subject to vertical leads finding in a beam mechanism. The generative structure of the included in the induction tantive, it is known as "necond order planic analysis".

For steel structures, the deformation beyond initial yield depends upon the

41

con-section classification. For example the CSN: [Canadian lumitate of Stef Contraction] classifies need section into fore categories. "Class 1" refer election permits they latestic happed overlappent and increased nations facilitating stress redstribution to other locations after the formation of initial plattice large at a critical location. The moment-carvateur reliationship is "statical" as represented by the bold line in Fig. 3.4. "Class 2" need sections after that, Classes 2 and 4 do not permit shift plattice for large notations after that. Classes 2 and 4 do not permit that plantification of the crass-sections. Therefore, all stot frames that are designed with class 1 accions and new redundances (tatical inflateriminacy), will have the strength beyond there music institutes the two reductivity of the strengthetions.

As long as the material has ducille behaviour, direct secant method can be used for analyzing the limit load in the similar way. In concrete structures, beam then have reinforcements below the "balaxee" limit (as is almost always the case) are known to exhibit ducille behaviour similar to that shown in Fig. 3. 4 by the bold line. For all who cases the stress reinforbload increment concrete of the firmt.

3.3 Alternative Methods

Since the compare implementation of the traditional parts: https://method.is combersone, it did not find practical utility. The method also ignores the effector of youry on the column means an mentioned above and boths to what is usually called as "tigid planticity". On the other hand, full scale nonlinear analysis of practical itself or concrete finance is even more involved in terms of effects and the erect to why should be apprecised.
on a mutic bain. Some of these difficulties can be overcome through the development of alternative methods. Such methods need to be simple, robut and methods. These alternative methods can be extensions of the currently available methods. One such procedure which is impired by the traditional plants: hingr analysis [Nod. 1997] as well as the robust method plants. [1997] is described and implemented in the present depice.

3.4 Inspirations from Plastic Hinge and R-Node Analysis

In the section below, the main inspirations for the development of the method based on direct secant modifications are described [Adluri, 1999 & 2001, Bolar and Adluri, 2001]. The points also include some of the main assumptions.

- After the first yield attest takes place, structural members begin to lose their stiffness, gradually becoming members with negligible tangent stiffness. The secant stiffness represented by the slope of the line joining the origin to the current point also-changes as the load increment.
- 2. The distribution of moments and other stress resultants are to be determined by the relative stiffness of structural elements. The changing of the stiffness of certain parts of structural elements can load to the changing of these stiffness, if Ausona, will give the "years" secant stiffness, if Ausona, will give the "years" secant stiffness.
- There are three types of plastic failure: complete collapse, partial-collapse and over-complete collapse. Complete collapse and over-complete collapse are achieved by removing the redundancies until structures become determinate.

Partial collapse happens when part of the structures is determinate. The changing of indeterminate structures to determinate structures is achieved by reducing the redundancies through the progressive formation plastic hinges.

- 4. At failure, the moments at plastic hinge locations are all equal to the plastic moment M_P of the respective sections. The section behaviour is assumed to be "ductile" where some transition zone might be present as well.
- 5. The scalar method is based on the understanding that the terms from the first decise analysis on the work of works in the structure. Thus, there will be a different search modular for each point in the structure. Thus, there will be a different search modular for each point adong the dippl of the acceleration as works. The structure is the structure is the structure of the monthers. The interface module point is the structure is the structure is the modular for data by the first structure is the structure is also based on a similar encourse. Instance is of modifying the modules of encoding in the module modules is the structure in the structure is the str
- 6. In the R-Node method, the concepts of load and displatement control are used to find the limit load. In the GLOSS diagram, the following up angle θ is equal to 90⁴ at the time of collapse. Similar concepts are applied in the Direct Secant Method.
- 7. In the t-node method, the final limit load is achieved by increasing the average peak stress to the yield stress while increasing the applied load by the same propertion. The analogy of two-but model is used for this purpore. The necessarionality factor denotes the limit load factor. Similar concepts are used by

the Direct Secant Method.

 Although the R-Node method can not be applied to determine the large deflection or stability of a structure, the method can be used as an inspiration for further develoement of the Direct Secant Method towards these goals.

These impringes from thatford prints: hinge analysis method and the R-Kolos method have been used to enhance the methods loading to the "Deters Starth Method". The methods have previously been described by Adult (1999, 2011). Bolie & Adult (2006) and it to obtain the limit load of various plate metasers. They utilize method generatively services and during analysis to estimative the ellissipa starts of plates. In the present work, this method has been used to obtain collepse taked f metasers and to analy the adulty and large defections ellissifies of themes and some. The fultrowise starks describes present and limits analysis of littery of the complex.

3.5 The Direct Secant Method

Addin (1999) proposed a method called "Direct Secant Method". It is inspired by the existing methods to find the limit hand using simple charts multipues --mainly the Kohole family chroholess. In this method, approxylamic analysis in first carried on without the considerations of yield stress. Based on the result: calculated from the first charts analysis, the rigidity in modified, the second stress the processes in a dampet at calculated and the second stress stress and the size of the second stress. The modified structure is malyred again with the same band, the same regress that the new rightly. Addre the second calculation and the same regress that the new rightly. Addre the second again with the same band, the same regress that the new rightly. Addre the second stress end stress stres These moments (or stress resultants) may have nothing to do with the corresponding the peak values from the first analysis. These peak values can be used to predict the collapse mechanism and the limit load. This technique gives a very good estimation of the limit load. The basis negata and principles are illustrated below:

 A purely elastic linear analysis is first carried out. In the analysis, stability is not under consideration. Global er local collapse due to platitisty and buckling never happens in this analysis. By using FAA, the bending moment at any location can be easily obtained. The first elastic analysis is of significance for investigating the distribution of bending moments.

 The results from the first purely elastic analysis can be used to modify the rigidities of the structure to be analyzed. Adluri [1999] modified the moment of inertia of the beam based on the bending moments.

$$I_{aau} = \frac{1}{|M_{*}(x)|}I_{abt}$$
 (3-8)

The proportionality constant can be chosen to be any nonzero value. To avoid numerical difficulty, Adluri [1999] used the maximum bending moment as the proportionality constant.

$$I_{new}(x) = \frac{M_{0.mm}}{M_{0}(x)}^{0} I_{nil}(x)$$
(3-9)

Usually, the exponent q can be taken between 1 and 2. In his analysis, he took 1 as the exponent. If the Young's modulus is kept constant, from this equation we can find that the modification stiffens the flexural rigidities of all the elements of the structure. Because M_{n-1} is not less than M(x), $I_{-1}(x)$ is greater or equal to $I_{n+1}(x)$. However the significance of the modification is not like this. As mentioned earlier in this charter, the distribution of bendine moments does not simply depend on the absolute flexural rigidities, but on the relatively flexural rigidities. This modification also changes the relative flexural rigidities of structures. Because Minum is a constant value, the elements which have relatively areater bending moment will have relatively less flexural rigidities after modification. The element at the maximum bending moment location will have the minimum relative rigidities, and vice versa, Therefore, the modification implies that in the first elastic analysis, the rigidities of the elements that have relatively greater bending moment are softened and stiffened rigidities are given to the elements which have the relatively less bending moments. The modification actually means "harmony". The elements which are weaker when resisting the bending moment are given stronger abilities and the reduced abilities are applied on the stronger elements. Therefore, the modification can be understood as: The elements of the structure are given the same abilities to resist the bending moment. In other words, the elements of structure have the same orecortanities to reach the same bending moment. Of course, this same bending moment can be the plastic moment of cross-sections.

Note that the significance of the modification mentioned above is under the condition that the structure is of the same size of cross-sections. All the elements of the structure have the same cross-section so that they can have the same moment of inertia and the same plastic moment. If the cross-sections are not the same between each element, such as if the beams are stronger than the columns, then the "harmony" cannot be simply achieved in this way. This will be mentioned later in this chapter.

3. The second static analysis is exceeded on the basis of the moliformit in tory 2, by using FAA, the distribution of broding memory can be some elements whose moments are generer than the error of neighboring densers. These elements is the broding memory diagram are of peak moments, but how the moliformit of moments are generated in the moliformit elements, which means albungh the molification of rightform gives all the elements the same elements that are moliformit elements, which means albungh the molification of rightform gives all the elements the same experimention to reach the plank memory the possible locations of plants lingers. The posk memorits but there are considered as the potential locations of plants lingers. As mentioned above in this shaper, the plants context persons through both the both the plant of the potential locations of plants lingers. Note that the same energit plant is analyzed on the plants memory. Thus, locating the plants linger both the same energit plant is allower in this shaper, the plants contingenergite the plants lingers. These, bottom plants plants plants plants are samelined above in this shaper, the plants contingenergite the plants lingers. These, bottom plants the plants lingers. These locations are outplanted for determining the thres lingers. Using the plant lingers linger linger linger lingers. These, locating the plants lingers. These locations are outplanted for determining the thres. linger linger linger lingers. The locating the plants lingers. Simplers lingers lingers lingers. The locating the plants lingers. Simplers lingers lingers lingers lingers. Simplers lingers lingers lingers. Simplers lingers lingers lingers. Simplers lingers lingers lingers lingers lingers. Simplers lingers lingers lingers lingers lingers. Simplers lingers lingers lingers lingers lingers lingers lingers lingers. Simplers lingers lin

In the first quarks analysis, the maximum beading areas location in the first location of a planck hings. After the formalist of the first planck hings, the stress will be distributed. The other paper and the first planck hings, the stress will the locations of potential planck hings. In the second analysis, the modification of memore of merita is based on the beading memore distribution from the first analysis. The denomine are modified to have the man exportantize of restricts of plancting the spetial based to have exceeding the distribution of plancting the proteind bactions of plancting the same exceeding the difference have the distribution of planct hinges appearing at the same time. This explains why all the locations of peak moments in the second analysis are considered as the potential plastic moments.

Potential locations of plastic hinges are not the exact locations of plastic hinges contributing to the final collapse. So we need to select some of them to form the collapse mechanism. The selection of potential location of plastic hinges can be done in several ways. This will be littured in the three section of this chapter.

 After selecting the locations of plastic hinges, limit load can be calculated by the following equation:

$$\frac{P_{im}}{P} = \frac{M_p}{M_{put-m}}$$
(3-10)

where P is the externally applied vector load, and can have any non-zero value. M_{p} is the plastic moment, and it is determined by the geometrical properties of the remov-section. $M_{pole-arc}$ is the peak secregal bending moment in the second analysis. At the locations of these peak heating moments, plastic hinges form and they combine the plastic college.

In the R-Node method, the limit load obtained in the R-Node method depends on the load factor that is equal to the ratio of the yield stress and the peak average stresses.

$$P_{im} = \left[\frac{\sigma_{j}}{(\sigma_{c})_{r-min}}\right]P$$
(3)

To better understand the significance of this load factor, we can modify the equation into:

$$\frac{P_{bm}}{P} = \left[\frac{\sigma_y}{(\sigma_z)_{r-mak}}\right]$$

(3-12)

3.6 Plastic Limit Load of Non-uniform Structures

Just piece spheric collapse, the factored load becomes the limit bolt and the trensers of earths hardinness are spin to the yield stress. The fraction, the numericants at both sides of the spatiations are high projection just piece to collapse. They are in the same state, which is called the "tollapse mate", at the locations of the documinator, $P_{\rm tol}$ the summity applied load and $(\sigma_{\rm tot})_{-\rm cont}$ is the reference tensors reading from the load P. Thereby they are also conceptualing, and they are also in the same state, which is called the "meaning and". The load factor axiatily is the ratio of properties in these two states.

In the R-Node method, the properties are the bending stresses. In the concept of dimension,

and load thus is proportional to the stress. Area is the proportional factor and is never changed in the analysis. Therefore, the exponent of the ratio of these two stresses is cannot to one. In canation (3-12), $\sigma \ \beta \sigma$) ..., is considered as the load factor.

We also can rewrite equation (3-10) into

$$\lambda = \frac{P_{im}}{P} = \frac{M_P}{M_{met-m}}$$
(3-14)

Where λ is the load factor. Similar to the situation of the R-Node method, $M_{\mu}/M_{\mu\mu\mu}$, and he viewed as the load factor. The numerators at both sides of the equation are in the same state of "collapse state" and the denominators are in the "normal state". In the concert of dimension:

and the length is not changed in the modification of the second analysis. Therefore, the exponent of the ratio of these two moments is equal to 1.

The total procedure can be described in Fig. 3.4. Line OA mergenesis field from parado-educit analysis with the drope $L_{2,0}^{-1}$. Because it is the relationship between the moment and the currenter, the sheep is the rightly. The moment line OA does not involve the limitation, so the moment can be higher or lower than the planter of the limit densities of the moment. The densities of the densities of the moment can be higher to else the the planter of the limit densities of the densities of the densities of the moment can be higher to else the densities of the densities of the moment can be higher to else the densities of t

following section, the line is expected to go down and intersect the theoretical line OB at the location C. The line AC goes downward directly intersect the X axis at the point D. A new slope is obtained based on the old slope.

Adluri [1999] obtained the new slope and the new secant rigidity is expressed as:

$$EI_{nur} = \frac{M_1}{I_{100}}$$
(3-16)

If the structure has geometrical nonlinear properties, the Yong's Modulus on both sides of equation (3-16) can be cancelled. Then the moment of inertia is modified base upon the equation:

$$EI_{ass} = \frac{M_p}{L_{100}}$$
(3-17)

where q=1 (duti,1999). Is fact, this is the modification that modifies the relative moment of instrins of attractures. The planic moment M_{c}/is is a constant value that does not infrance the modification of the relative moment of instrins of structures. Therefore, the proportionality factor can be used any momen value. Another analysis is done with the new slope, hemotive analyses are carried out util convergence occurs. This is represented by the carrie law A.H. if the line A.D der or go donessed, 2 cm go between the betwinstal line and vertical line, and $d^{2} \leq p \leq 50^{2}$. At this time, q is not equal to 1, and it can be other values. No matter low mach q is chosen, the intrarise calculations are performed utilit convergence homess. The members of the structure metilised caller are of the same cross-sections. If the yield attests is constant, the same cross-sections will lead to the same moments of indication and the same plantic moments. However, in the next caller, the corresponding of the structural members are not always the mass. It implies that the silutation of structural members with different plantic moments seeds to be included. If cross-sections are not be same, for solving non-sufficient structure, the technique of Reduce Mediated Mediated Structure ones.

The main principle of the direct scars meshod is that: all the elements are given besure addition to resist the bending movement and they are than given the same and the same meanset, and it can be the planfor movement. After the analysis based on the modification, see seends to find which elements relatively brunch the peak meanest. If the cross-scarims are not the same, the experimitivic given for each element are different. This is because the same experimenticies are given based on the modification which returbs from two factors: moment of intrinand planfor movement.

For the uniform structures mentioned earlier, the modification is given by the equation

$$I_{new} = \frac{1}{|M_{s}(x)|}I_{str}$$
(3.18)

The distribution of moments $M_{\eta}(x)$ is determined by the relative moment of inertia in the first analysis. The proportionality factor can be any nonzero value. From canation (3–18), the plastic moment can not be seen. This is because our purpose is to modify the relative moment of inertia but not the absolute values, and the structures are expected to reach the same moment—plattic moments, there is no need to give other proportional to factors instead of one. In fact the plattic moment can be put in the summerster:

$$I_{nov} = \frac{M_{p}}{|M_{n}(x)|} I_{obt}$$
(3-19)

This will not influence the result because the relative moments are not changed.

However, if the structural members have different cross-sections, the situation is different. With different plastic moments between each member, structures are expected to reach different values after the modification. To satisfy the purpose of "the same opperation", we should adjust the modification by

$$I_{new} = \frac{M_{\pm}}{|M_{\pm}(x)|}I_{au} \qquad (3-20)$$

where M_{ij} are the placic moments of cross-sections. The bending moment $M_{ij}(x)$ should correspond to the planet moment. In equation (2-20), the planet moments M_{ij} in to actuate where all should move models that the equation (2-20), the planet calculated from the first analysis $M_{ij}(x)$. After the modification, the structural number are of the same alloying the moments and thus of the same reportantiles to reach the same moments.

Finally, the load factor is calculated by the peak average ratio of the plastic and the moment after the first analysis.

$$\lambda = \frac{P_{in}}{P} = \frac{1}{\pi} \frac{1}{\sum_{i=1}^{n} \frac{M_{pattern}}{M_i}}$$
(3-21)

Therefore, the procedures for analyzing non-uniform structures are illustrated below:

- The first elastic analysis is carried out based on the original geometrical properties.
- 2. Modify the cross-section geometrical properties by using equation (3-20).
- 3. The second analysis is executed on the basis of the modification.
- 4. In the bending moment diagram of the second analysis, peak moments are considered as the potential locations of plastic hinges. Some of the peak moments are selected to form the collapse mechanism.
- 5. The load factor is calculated from the expression:

$$\lambda = \frac{P_{\text{int}}}{P} = \frac{1}{n} \frac{1}{\sum_{r=1}^{n} \frac{M_{r,rest-ser}}{M_{P_r}}}$$
(3-22)

For uniform structures, M_{p_1} is constant, and if it is equal to M_p , the equation can be simplified as:

$$\lambda = \frac{P_{sm}}{P} = \frac{1}{n} \frac{1}{\sum_{\substack{s=0\\s=0}}^{n} M_{s-smk-sm}} = \frac{1}{n} \frac{M_{s}}{\sum_{s=0}^{n} M_{s-smk-sm}}$$
(3-23)

This is the same as equation (3-14) mentioned above for the uniform structures.

Two factors can determine the plants moments: one in the yield transph and the other is the section modulus realing from the coss-section properties. The analysis methods also easily liakeds the consideration of different cross-sections but without involving different yield strength. However, if more than one type of material is used, the yield strength will be different. This is common in the real contractions of streatures. Therefore, the Robot Method including changing yield strength thold les proceed in factor work.

A non-million point fines is inoriging. The instantial methods are of offeness cross-sections with the same strength. This is shown in Fig. 316. The finate entrophetics is latered there shows in a same production, highly to a vertical finee that results in the beam mechanism and arbitret to bed of them that result is a combine modulum. There causes of non-mailtenn point finance are maded: the beam is strenger than to enclosure, beam point of them that results is strenger than to enclosure the strength with the beam in the beam is strenger than to one column and weaker than the chem is moments of instant are different. These moments of interair argin can delay are appelled to the strength methods required by the the beam-field wavely from plantic analysis. Load factors are used an the results.

3.7 Comparison between Direct Secant Method and R-Node Method

3.7.1 Modification

Direct scenar method: The modification is based speen changing the rightitist of areatures. The rightitist of structures include cross-scening generatival properties and material properties. This implies two advantages from can successfully modily the rightitist of attention. This implies two advantages from, is the competer program, engineers can advantage properties they within bondily, so the limitation of the competer programs on the workds. The consequence prolocation advantage of the rightitist are seen on the structure attention, the rightitist are seen in Section, the rightitist are of contrast, The new rightitist can be modified based on the all rightitist excepts at the location of hisper. The disadvantage is that at the relightin based based model. It is subscent, the moments at these baselines are the metation of hisper. The direct results at the baseline of a section of a section of the rightitist provide the rightitist are the function of hisper. The direct terms methods, terg the baseline of a section. Therefore, the pro-rightitist produces the function of a section of a section of the direct terms methods, terg the produces the terms of the regular terms. The function of the section of a section of a section of the direct terms methods, terg the produces the term of the section the rightitist are the out-function.

Rivide method: The modification is based on shanging the Voney's modulus of structural cross-sections. In the Rivide method, one has to molify the Yoney's Modulus without any other devices. This growthese difficulties for events FLA software. Also, the modification is based on equation (213), and the equivalent strenues of each cross-section σ_{e} may be equal to zero at cortain bottlems of cross-section. Numerical afficiations are then necessation: If here if we can successfully modify vary housday has the trans-sections, after here and house of each error-section, we till can not give accurate modification. In FAA, the changing of stress for each element mide the environment in not continous. Mooph number airor delements are given afference till each and the time of calculation becomes langer. The R-bode method also can not deal with the situation of the lange in trues structures. At the locations of langes, no bending moment lands to so uses. Therefore, the modification to meaning.

3.7.2 Element Type

The direct secant method uses member level demonts, in this case, beam demonts to model the framess. The direct secant modification is for the entire cross-section and depends on stress resultants such an moments and shears and does not directly depend on stress list.¹². As opposed to this, the R-Node method uses solid elements since the modification in based on local stress.

3.7.3 Failure mode and yield criterion

Direct secant method: Many failure modes now have been successfully solved by the direct secant method, such as plastic collapse and buckling. Any yield criterion can be taken by this method.

R-Node method: It can solve the plastic collapse. Until now, there has been no other extension for the limit load calculation. The R-Node method is now only used for a single vield criterion. New techniques for this method need to be developed.

63

3.7.4 Limit load

Direct secart method: Limit load is calculated by calculating the peak average momenter. The collapse mechanism can be determined by this method. R-Node method: Limit load is determined by the yield stress. Collapse mechanism can not be determined by this method.

3.8 Test Cases and Examples

The method described above using direct secant modification is applied to several frame type structures to predict their limit loads. The examples include portal frames, multi-bay and multi-stoney frames with different loadines and arrangements. The loads include concentrated and distributed forces, vertical and sway forces. Three methods are used to find the limit loads and compare the results, viz., traditional plastic hinge analysis (as described by Neal [1977]), full scale nonlinear analysis using FEA and the direct secant method presented in the current chapter. It must be noted that the R-Node method [Seshadri, 1977], which forms part of the inspiration for the current work gives very close results to that given by the direct secant method albeit with significantly greater extra effort (both programming and computational). For comparison purposes, the FEA nonlinear analysis is taken as the base instead of the plastic hinge analysis. This is so since the FEA nonlinear analysis can include the effect of suaw on column moments where as the plastic hinge analysis ignores it. Besides, the plastic hinge analysis is not easy to apply for multistory frames with mixed loading conditions. Finite element nonlinear analysis has no such limitations.

3.8.1 Finite Element Modeling

For the purpose of analysis the structures have been modeled using the finite element software AASYS [2006]. This was used for both the nonlinear analysis and the Direct Secant Method. Traditional chartic analysis using FEA is well known [e.g., Logan, 2005] and is not be explained here in detail.

The direct sector analysis used in the current chapter has been implemented in ANSYS using beam chemican (IREAM). As nontrioned above, this facilitates the modification of the sector trigidity at the local of the entire conversarios. The modification is based on stream remainst make an anoments and is not based on the single analysis in earlier are obtained, the modifications to the digitily (in this case, to the moment of tention of earlier based models and the case, to the moment of particul of earlier based models and APGU, resolves which NASYS. Sample contacts the modifications are given in APGU, resolves which NASYS. Sample contacts the modifications are given in an and the analoge of the fiber. They are based on the samfer tank modifications implemented by Bolar and Athui (2006) sho did the same for park treatment.

For the notinear analysis, the structures were modeled using different options available in ANSTS (for comparison purposes and conventione). The nonlinear analyses have been carried out using beam dementin BEAML in each case, much convergence studies have been carried out. The results for each element type have been tested with hower theoretical results for test cases.

Several test cases are shown in Figs. 3.5.1 to 3.16.1. In the figures, the properties shown next to the figures are all non-dimensional. The lengths, loads, etc., are all the ratios of a corresponding base unit of value 1. Normally, load factors are greater than 1 in the real cases. In this work, the loads were chosen such that the load factors are less than 1 to facilitate the work of FEA nonlinear analyses using a single load step.

3.8.2 The Description of the Test Cases and Examples

3.8.2.1 Uniform Frames

The cases tested by FEA nonlinear analysis and the direct secant method include: beam mechanism, sway mechanism, and combine mechanism as well as several relatively complicated structures subject to concentrated forces and distributed forces.

Table 31 shows the results of the funces presented in Figs. 35.1 to 33.1 to 30.9 to by the direct scoutt method. The magnitude of the equite backing moments in the memory on of the direct scoutt method are listed in the table. By using the results in table 31, the limit loads in terms of load factors analyzed by the traditional plotte analysis (except researce in Figs. 3144 to 33.5) and full scotts molliner analysis. The results obtained by two methods in terms of load factors are also there in the limit loads in terms of loads are studied by the traditional plotte analysis (except researce in Figs. 3144 to 33.5) and full scotts molliner methods. The frames observes in Figs. 314.1 and Figs. 315.1 are related to analysis (except the direct and analysis) and figs. 315.1 are related to an advance and the limit of the limit of the frames of the limit bold for there an advance. The direct method is used to analyse the limit bold for them an advance analysis in darks limit of the limit frames of the limit bold for them and the limit of the direct scott method is used to analyse the limit bold for them and them of the limit of limit of limit of limit of limits of limits

In table 3.1.1, the results of FEA nonlinear analysis are used to compare with the results of both the traditional plastic analysis and the direct secant method. It is assumed that if the results of FEA nonlinear analysis are closed to the ones of traditional plastic analysis, the results of FEA nonlinear mathysis is considered reliable and can be used to compare with the direct secant method. In the case of relatively complicated structures, traditional plastic analysis seems tedious and sometimes is not feasible. FEA nonlinear analysis in this case is used to be the only base for comparison. In table 3.1.1, the difference of the comparisons is within 3.3%. It verifies that the results of the direct secant method and acceptible and relatible.

Fig.3.51 presents a portal frame subject to a concentrated force in the middle of the beam. It is used to test the beam mechanism, Fig. 3.5.2 and Fig.3.53 presents the beening moment distribution of nonlinear analysis and the distribution of reanalysis in direct secant method. In Fig. 3.61, newsy mechanism is tested. Fig. 3.7.1 shows the test of combine mechanism.

A multi-lay structure, which is subject to vertical and latently the loss manyor by Noai (1977). To obtain the limit load, the firsters are applied independently and each conseponding collapse mechanisms are combined projectively to other instimutes late new which is the limit load factor. The same procedure is followed by the analysis of the first secaration works. From Fig. 3.11 to Fig. 3.111, all the independent and combined mechanisms are tend by FDA structures and the structure mechanism during the structure of the structure mechanism obtained in the combine mechanism during Fig. 3.11, and the limit load factor in this const 10.571.

Fig. 3.14.1 presents the text of the partial collapse of a frame. The collapse mechanism is the beam mechanism. The plasmic hinge locations are marked in Fig.3.14.3. Fig.3.15.1 shows a more complicated frame subject to lateral usiformly distributed forces and vertical econectimated forces. The collapse mechanism obtained

67

is the sway mechanism on the first storey of the frame. Ten plastic hinge locations are marked in Fig. 3.15.3.

3.8.2.2 Non-uniform Frames

A petid fame with non-uniform generation properties subject to a hare force and a vertical force in shown in Fig. 3.3.6.1. The limit hand prediction is performed by the effect scears method and compared with FLA molecules analysis. These carso including different properties of burns and columns are given. For each care, three basic collipses mechanismic (beam, you and conthism mechanisms) are multiput for HEA southness analysis and the direct secars analysis are presented in tables 3.22 to 3.42. The direct secars muchoid given good estimation and the difference is with 70.

		8	9	0	0	*	0	0	0	Location
							36265	43735	36265	Fig. 3.5.1
						76219	67771	67780	76230	Fig. 3.6.1
						18919	79014	65213	80627	Fig. 3.7.1
							37876	30455	25214	Fig. 3.8.1
							30257	36545	40651	Fig. 39.1
				29781	34652	36478	32914	29764	34632	Fig. 3.101
				33719	41274	38409	4265	42612	38284	Fig. 3.11.1
			33891	XXX	1000	25190	39482	334-8	20886	Fig.3.12.1
			39045	40012	41000	39515	52542	40607	26680	Fig. 3.131
							0,1281	88920	95116	Fig. 3141
8.33E-65	1402E+05	1420E+05	1424E+05	\$0:00.8	\$0(3973	8.562-05	8,74E+05	\$.67E+05	\$0+325+6	Fig. 3.151







Fir. 3.2 Sway Mechanism for a Portal Frame







Fig. 3.4 Direct Secant Estimation of Plastic Limit Load [Adluri, 1999]











Fig. 3.5.3 Direct Secant Method Results from Reanalysis

for the Problem in Fig. 3.5.1







Fig. 3.6.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.6.1



Fig. 3.7.1 Portal Frame Subject to Both Lateral and Vertical Forces



Fig. 3.7.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.7.1















Fig. 3.9.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.9.1































Fig. 3.13.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.13.1







Fig. 3.14.2 Nonlinear Analysis Results for the Problem in Fig. 3.14.1



Fig. 3.14.3 Direct Secant Results from Reanalysis for the Problem in Fig.

3.14.1






Fig. 3.15.2 Direct Secant Results from Reanalysis for the Problem in Fig. 3.15.1





Table 3.2	Load Factors 1	for the problem	in the Fig. 3	.16.1

	Direct secant method	Nonlinear Analysis	Error
Beam Mechanism	0.792	0.812	2.46%
Sway Mechanism	0.316	0.316	0%
Combined Mechanism	0.432	0.436	0.92%









for Case 2

	Direct secant method	Nonlinear Analysis	Error
Beam Mechanism	0.420	0.416	0.96%
Sway Mechanism	0.804	0.828	2.89%
Combined Mechanism	0.552	0.536	2.99%

Table 3.3 Load Factors for the problem in the Fig. 3.16.2

Case 3: Beam is stronger than one column and weaker than the other





for Case 3

	Direct secant method	Nonlinear Analysis	Error
Beam Mechanism	0.980	1.004	2.39%
Sway Mechanism	0.804	0.816	1.47%
Combined Mechanism	0.744	0.764	2.62%

Table 3.4 Load Factors for the problem in the Fir, 3,16.3

Chapter 4

Estimations for Frame Stability

4.1 Introduction

Stability in a major doing nominering for structurel obtainers, finance and additional structures adaptive to compression or compression and entering in some constrainties. The classic analysis of entering in the second structure of a significant (see, e.g., Bladder (2004)). Second of a invivational foregore 2. The detoreal bases for adaptive spars finand as early and the second structure of the second structur

In this chapter, these routes are bypansed in favour of a simple attempt to explore the possibility of obtaining the critical load capacities for fame stability by using algorithms based on methods in Chapter 3. In most be clearly noted that this is only a radimentary attempt and hence theoretical right for full investigation is beyond its store.

4.2 Observations from Elastic Buckling Theory

We can gather several observations from the theory of elastic buckling that are useful for the present study.

- 1. If a performant performance in supplies a small consention and leves, it will have only a small compressive displacement along the column reading straight. The column relations is not state equilibrium. Lateral displacement examed by a situating lateral first; (or moment) at this state will studies here the distribution factor is momented at the column state state and lateral distribution factors. These will readed the state of the state
- 2. One of the key observations from clustes buckling analysis in that the Brenn at a fifthere that restits benefing a pregnetively workness with the increase in the clustes. The stiffness contrainly readers negligible levels when the staid force in the clustes. The stiffness contrained state of stiffness may be simulated to some cluster using least. It is postimely that this is not cluster and the clustes at the cluster at these cluster at the clust
- 3. The same observations as above apply to columns that are part of a frame. Such frames will also have "buckling" loads the same way that columns do. This is the case whether the frames remnit or reevent side sway. These frames can have

different types of connections between beams and columns. The column stability will depend upon the effective end restraint. The beams and their connections will act as partial end restraints to the columns thus modifying the effective length of the columns and their buckling capacities.

- 4. For frames, the lateral deflection in columns may be produced by axial forcer, lateral forcers and/er bending moments. We can call the deflection easied by lateral forcers and direct bending moments as the primary deflection while the deflection produced by axial forces as the secondary deflection. This "secondary" deflection in fact in mere splittlength of the current chapter.
- Fig. 4.1 shows the diagram of a segment of an isolated column. If we take the moment equilibrium,

 $Qdx + Pdy + M - (M + \frac{dM}{dx}dx) = 0$ $Q = \frac{dM}{dx} - P\frac{dy}{dx}$

(4-1)

If the lateral force O is ignored,

$$M = Pdy$$
 (4-2)

Therefore, if lateral force effects are removed, the moment increment in the column is equal to the product of axial force and displacement increment at that location.

6. The lateral forces such as Q in Fig. 4.1 cause lateral displacements. The displacement patterns due to these loads may not be the same as the displacement

patterns at the time of column or frame buckling. Therefore, the lateral loads or moments may have the max displacement at one location while the buckling effect may have the maximum displacement at a different location.

7. Just prive backling, the maximum diplacement of each column one to considered to be the critical displacement. This implies that if any displacement concereds the critical displacement, failure and to each owner interace that an mentioned above, when backling sectors, the huge displacement location is not easily the same as the critical displacement location. However, these displacements are most likely to be close to and other. However, these manyback of other acate methods, while will be discussed likes.

4.3 Inspiration from Robust Plastic Limit Load Analysis

In addition to the observations above, we can use some of the ideas gained in Chapter 3 to try and simulate frame buckling.

1. We still assume that the frame bockling for this situation does not follow the classical mode of "bifurcation" or a sudden classical pair is status. Rather, the backling is a transition from more or loss linear denis dedoemation (insubling both axial and lateral displacements of members) is a relatively very large set of lateral displacements of or some himportures are the fast that its holds place at displacements in or of a much importune as the fast that its holds place at This is somewhat similar to the fast that at plantic limit hands, the curvature of the beam at the lateral at the similar to the fast that at plantic limit hands, the curvature of the beam at the similar to the fast that at plantic limit hands, the curvature of the beam at the similar to the fast that at plantic limit hands, the curvature of the beam at the similar to the fast that at plantic limit hands, the curvature of the beam at the similar to the fast that at plantic limit hands, the curvature of the beam at the similar to the similar based. The magnitude is not important status that the similar to the similar based of the similar based of the similar based.

- 2. The relicionship between applied lawle and the displacement at critical leadents' in summediat analgous to that between measures and curvature of the critical convesceion in Chapter 3. Both the relationships have initially linearly increasing performs which must targe holostatile performs the linear leadent of the linear leadent of the strength of the streng
- 3. The plantic limit hole estimation of Captur 3 relies on a secant molfication of the eighty to simulate the reductivities of releasing movement cannel by the enset of yielding. The bending moment re-distribution reflects on the relative secant nighty of affinests parts of the finame. Similarly, we can promider that the abality of affinests parts of the finame. Similarly, we can promider that the Chapter 3, the secant modification here oney and gives all the parts of the finame one quart "separativity" to buckle. Schematically, this modification is shown in Fig. 4.2.

It must be noted that the word "secant stiffness" is used loosely here. It is not exactly the same as that used in iterative modified Newton schemes for nonlinear matrix analysis.

4. Just as we obtain locations of peak redistributed moments in Chapter 3, we can try to obtain locations of peak displacements in the structure whose stiffness is modified by the scheme shown in Fig. 4.2. These peak displacements can potentially be indicative of frame instability.

4.4 Robust "Secant" Analysis Trials for Elastic Buckling

Using the observations above, we can attempt to investigate the elastic buckling of frames. We repeat that this is a very preliminary idea and needs to be thoroughly analyzed before any further development.

It is postulated that frames exhibit "buckling" behaviour when a certain critical displacement from simple elastic analysis exceeds a certain limit.

In the following, <u>empirical procedures</u> are proposed for estimating the critical load for a column member in a frame.

- 1. For set patters of applied loading <2, M2 and an athress load multiplication factor A, we carry not an initial chaics analysis. This gives us the general hyperchemoty pattern for the structure which we assume increases monotonically with load multiplication factor antil one or noves members or the frame as a whole decides us backle. An americaned artifice, over at backling, the displacement jump at assumed to "bifurdar". In athre increases significantly at some points than presenting inferto-possible y discover interplacement in an antiplication factor.
- 2. In Fig. 42, the line OA represents the linear dustic relationship between applied load and displacement. It is obtained based upon the initial utilitiess (obsyc) Ki. As the load increases, the atliftees remains relatively constant while the corresponding displacement increases proportionally. This continues util a critical displacement d_i is reached at point A. The tangen attiffees becomes

effectively very low thereafter. (We are, as mentioned above, neplecting a transition none near point A). Let us assume that the load applied in F, and that it produces a displacement d₁ based on initial stiffness K₁. In order to simulate backling, we can modify the stiffness of the segment using the secant line shown in Fig. 4.2 (similar to hus of in Cherser 3).

$$I_{new} = \frac{d_{ab}}{d}I_{ad}$$
(4-3)

where, Is the moment of inertia for the segment under consideration. The value of d_{ab} is used alone the value of d_{ab} is not actually known and since any reasonable where we are available of the second segment of infinite second second and that in Chapter 3. Note that we are modifying possess of inertia bidsh is proportional to the stiffness of the second secon

Re-analysis of the frame is performed on the basis of the modification above. The locations of peak displacements are obtained. These locations might have shifted from the corresponding ones in the previous analysis (analogous to the situation in Chereter 3).

3. After re-analysis, the critical load can be calculated by one of the following two <u>empirical methods</u>. The first method obtains the load factor corresponding to critical load from the ratio of two average displacements. The second method uses the ratio of moment integral and displacements integral. Both the methods are illustrated below:

4.4.1 Method 1 for Critical Loads

Let P_{arametric} be the studi compressive force in the column, M_{aramet} be the broking moment at the location of peak displacement and C be a factor depending on the tryp to frame. Compose an explaindor: displacement, data, ma blocks. There may be more than one peak locations. Note that the subscript "maye" is not to be confined with the multiplace constraint used in second commercial sub-tour packages such as ANSVS and AMADCS.

$$d_{apc} = C \frac{M_{dep-post}}{P_{contrast}}$$
(4-4)

The load factor λ for the critical load P_{ir} is obtained from the peak displacements,

$$\mathbf{P}_{ac} = \mathbf{P} J_{a} \quad \lambda = \frac{1}{n} \sum_{i=1}^{n} d_{spc} \\ \sum_{i=1}^{n} d_{soci} \quad (4-5)$$

where, d_{nut} is the peak displacement in the reanalysis.

The factor of depends on the type of frame irreportion of the attual physical property values. We do not yet have the necessary theoretical development to dottmine the functional property of the development of the theory. It has been decided to empirically estimate the factor using finite element generative sources analysis. The factor of his/bay to a fanction of centain key non-dimensional properties of the many. However, of the traject study at the same of the theory of the many stars and the same of the same of the theory of the same of the same. However, of the traject stars at the same of the theory of the same of the Fig. 4.3 shows results for the factors C for point frames with different space mixing where, J_{ij} is the length of exists and J_{ij} is the length of the beam. The maximum values of C is 0.817 when $\frac{J_{ij}}{J_{ij}} - 1$ and the minimum values of C is 0.4641 when $\frac{J_{ij}}{J_{ij}} > 1$. The curve shows limit difference when $\frac{J_{ij}}{J_{ij}} > 1$ and it form around a summat values in The curve shows limit difference when $\frac{J_{ij}}{J_{ij}} > 1$ and it form around a summat value in J_{ij} .

curve. This constant value is selected to be 0.72.

Fig. 6.4 does the comparison between the band factor λ for antiducur analysis what that for robust secant analysis when C=0.52. In the figure, beams are short beam that for robust secant analysis when C=0.52. In the figure, beams are short beam denotes robust neura analysis. In most be spinsed out that the factor C is being calibrated using the "eners" analysis with the expectation that such factors will be invariationed in further work and proceedings for finding them will be developed within recovers to the backet/labetium of them.

4.4.2 Method 2 for Critical Loads

In empirical method 2, after reanalysis, the critical load is calculated as below:

 Identify the locations of peak displacements and the nearby locations of zero displacements. Find the bending moments and displacements at these locations.

2. Calculate

 $\Delta M = M_{der} - M_{de}$ and $\Delta d = d_{er} - d_{de}$ (4-6)

where, $d_{s'}$ is the peak displacement in reanalysis, d_s is the zero displacement, $M_{ds'}$ is bending moment at the peak displacement location and M_s is the bending moment at zero displacement location.

3. Estimate the critical load by using the equation

$$P_{cr} = \beta \frac{\Delta M}{\Delta d}$$
, (4-7)

The factor β is analogous to the factor C in Method 1. It is calibrated using the same nonlinear analysis as used in Method 1.

Fig. 6.3 mentionlass the lates β . For point finness subject to a varied first on the beam, where, f_{ij} is the laugh of columns and f_{ij} is the length of beam. The maximum values of β is 22.33 for $L_{ij} < 40$ minimum values of β is 0.378 when $L_{ij}^{i} < 2$. The cores shows limit difference when $L_{ij}^{i} < -1$ and it front around a contast value. It implies that a contast value can be used to over the range $L_{ij} < -1$. The correct value is indexed to 10-22.

Fig. 6.8 shows the comparison of the load factor between generation of motifier and the strength of the strength of the strength of the strength of the the theoretical load factor for a partal fame with different aspect ratios. The dashed line dimetes the curve for subset strengt analysis when $\beta = 0.22$. These two lines are close and do an show much different websit $277 \le \frac{1}{L_{1}} \le 1$. The maximum error values in the form in VS.

4.4.3 Comparison Between Method 1 and Method 2

It is obvious that Fig. 4.3 and Fig.4.5, show general trends that are proportional to each other. The value $d_{\mu\mu\nu}$ in Method 1 is equal to Δd in Method 2. The value $M_{\mu\mu\nu}$ al Method 1 is proportional to Δd in Method 2 because $\Delta M = M_{\mu\mu\nu} - M_{\mu\nu}$. The difference between the factors ρ and the factor *C* comes from the difference between $M_{\mu\nu}$ and ΔM .

Although Method I and Method 2 are obviously closely related, the intent of AM in Method 2 in not the same as the value of M_{max} in Method 1. The see of AM was mean to represent the total areas of bending memory diagrams between the peak backeton and the zero boation than indicating "upping" per unit diplotment that causes backling. Since the use of an integral is somewhat cambersone at this stage, it was accided that a simple "diagrap of value" will be used intend. This needs function involutions.

It must also be pointed out that we used the factors C and β as constants (for the range of aspect ratios selected for the present study). They are not actually constants as can be seen from Figs. 4.3 and 4.5. In fact for $L_2 > L_3$, these factors seem to be finately changing. This needs for the constrainty and the moving of a point point.

4.5 Results

Several portal frame problems have been analyzed using empirical Methods 1 and 2 described above as can be seen from the points represented on Figs. 4.3-6. The results of some of those analyses are presented in Figs. 4.7 to 4.13. For the analyses, the following physical data was used (Table-A.1). Although the data was needed to be given to ANSYS in order to obtain numerical results, the data presented can easily be those to be not-dimensional in their nature.



P	28,000
L	1200
L	800
L ₅	400
b	8
h	10
E	200,000

The analyses include portal frame with one load, two loads and three loads on the beam as well as one lateral (horizontal) force parallel to the beam to simulate the side sway due to wind. The frames cover different aspect ratios.

As one besen from the results presented, the errors from the proposed analyses are reasonably small in comparison with the results of generative nonlinear analysis. It must be noted that the generative nonlinear analysis results are evolution of different from these in Figs. 43-6 since these use the matrix eigen value analysis. However the difference is within indenthe margins. Typical input files for ANSYS analyses are included in the Arependices.



















Fig. 4.6 Comparison between Geometrical Nonlinear Analysis

and Robust Secant Method for \$ =0.22











Fig. 4.8a Portal Frame with Concentrated Force on the Beam Case 2 (L2 = L1)

	Properties	
PL_1^2	4	<u>L</u>
ET	L1	L ₂
96	0.5	0.5

Result from Reanalysis

$\frac{d_{poll}}{L}$	$\frac{M_{peak}}{PL_{0}}$	
8.41e-5	2.08e-5	

Estimate of Critical Load

C	Geometrical Nonlinear Analysis	Robust Secant Load factor	error
0.72	0.2135	0.1776	16.8%



Fig. 4.8b Displacement Distribution in Reanalysis for the Portal Frame in Case2



Fig.4.9a Portal Frame with Concentrated Force on the Beam Case 3 (L2 > L1)

	Properties	
PL ²	<u>L</u> 2	<u>L</u>
EI	L,	L ₂
54	1.33	0.5
54	1.33	0.5

Result from	Reanalysis
$\frac{d_{poll}}{L_1}$	$\frac{M_{pul}}{PL_1}$
1.99e-4	8.92e-5

		_	

Entering of Children Frank				
C	Theoretical Load factor	Robust Secant Load factor	crror	
0.72	0.3372	0.3226	4.3%	





Portal Frame Case 4



Properties				
<u>L</u>	<u>L</u>			
4	L_2			
0.5	0.33			
	Properties <u> <u> </u> </u>			

	Result from Reanalysis	
peak displacement	1 diput	$O \frac{M_{past}}{PL_s}$
	3.89e-4	8.88c-4
peak displacement®	@ dma	$\otimes \frac{M_{pask}}{PL_{1}}$
	4.21e-3	1.58e-3

Estimate of Critical Load

C	Theoretical Load factor	Robust Secant Load factor	error
0.72	0.3707	0.3911	5.2%







		Properties		
$P_{1}L_{1}^{2}$	$P_{1}L_{1}^{2}$	L.	<u>L</u>	\underline{L}_{4}
EI	EI	I,	L	L_{2}
38.4	28.8	0.75	0.3	0.5

peak displacement	$\odot \frac{d_{pat}}{L}$	$\bigcirc \frac{M_{put}}{PL}$
_	3.89e-4	8.88c-4
peak displacement®	$\otimes \frac{d_{net}}{L}$	$\otimes \frac{M_{peak}}{PL_{p}}$
	4.21e-3	1.58e-3

Estimate of Critical Load

C	Theoretical Load factor	Robust Secant Load factor	error
0.72	0.2824	0.2858	1.2%









	Properties						
$P_1L_1^2$	$P_{2}L_{1}^{2}$	$P_{3}L_{1}^{2}$	<u>L</u>	<u>L</u>	<u>L</u> ₄	<u>L</u>	L
EI	EI	EI	4	L ₂	4	L_2	L ₂
15.36	11.52	9.22	0.63	0.3	0.2	0.3	0.2

Result from Reanalysis

peak displacement(1)	1 dpat	$\bigoplus \frac{M_{post}}{PL_s}$
	8.18e-3	6.85e-3
peak displacement2	@ dreat	$\otimes \frac{M_{post}}{PL}$
-	1.24e-2	7.82e-3

Estimate of Critical Load

С	Theoretical Load factor	Robust Secant Load factor	error
0.72	0.49	0.52	5.2%









	1107		1
$P_1L_1^*$	$P_2L_1^*$	4	5
EI	EI	L	L_2
38.4	9.6	1.5	0.5

· · · · · ·	 	-
KOUR	 	 агум

dput	Mpeak
4	PL
0.21	0.11

Estimate of Critical Load

C	Theoretical Load factor	Robust Secant Load factor	error
0.72	0.3247	0.3011	7.3%





Chapter 5

Estimation of Large Deflections in Beams

5.1 Introduction

The study of large deflections is a vast and complicated subject. It has many applications and is used directly and indirectly in several design situations. A very preliminary attempt at estimating large deflections of beams or beam type structures using inspirations from direct secant analysis is made in this chapter.

Tool differents in the vectors uses of the differentian in all discretions. There are many indicators that can be used to identify if the prediment of interest results large different conductations, for example, if the value of the notation of or bosons some the tenily replaced with sind or vice vecus. In a design situation, typically, a stiffaction in excess of "spars125" is considered according small alterbiotism line. When implicadiotections that prime, vanismation methods methods in the implicad positions. This deviation is considered to be larger smaph so that the equations found on the original generaty are no longer valid for a good estimate of the babarione.

In this chapter, the large deflection of beams is investigated using Euler-Bernoulli beam theory, linear and nonlinear FEA and robust secarat analysis used in earlier chapters. Three cases of beam, vize, cantilever beam, simply supported beam with a roler and simply supported beam without rollers.

5.2 Euler-Bernoulli Beam

5.2.1 Linear Theory

Demonstry (Judo-Bernsoffi Isan equation is well known and is widdy taught in engineering carriedum (see, e.g., Hibbler [2008]). The basic assumption is that the sum section that is origingly plass remains plass and resolution. The popularly known form of the equation assumes that the beam is instropic. It relates the applied loads and the directions caused by the hash. With reference us beams of the kind from Fig. 3.1, the equation is represented as:

$$\frac{d^2}{dx^2}(EI\frac{d^2y}{dx^2}) = w$$
, $EI\frac{d^2y}{dx^2} = M(x)$, $I = \int \int y^2 \cdot dy \cdot dz$ (5-1)

where, E is the material modulus, I is the moment of inertia of the cross section at any x_x , $y_x(x)$ is the the deflection at any point x_x M(x) is the bending moment, and $w_x(x)$ is the intensity of the distributed load at x_x . The equations are derived using the following assumptions:

 Beams have small deflections. The stresses, except bending stresses, are negligible. Beams behave elastically.

$$x = -\theta = -\frac{dy}{dx} = -\sin\theta \qquad (5-2)$$

2. The stress resultants Bending moment and Shear force are given by,

$$M(x) = \iint y \cdot \sigma(x, y) \cdot dy \cdot dz \qquad (5-3)$$
$$V(x) = \iint \sigma_{ay} \cdot (x, y) \cdot dy \cdot dz$$

(5-4)

Since the main assumption is that plane cross-sections remain plane after deformation, the shear effect on the deflection is not included.

5.2.2 Modified Euler-Bernoulli Equation

If beams experience large deflection, the linear beam theory illustrated above is not entirely correct since the magnitude of θ is no longer the same as $\frac{dy}{dx} = \sin \theta - \tan \theta$.

A satilater beam adject to a constraint of first of the first of the source of the first of the shown in Fig. 3.1. The versical for d^{2} ensures writed independent d_{1} , the initial of d of the beam. The total deflection d^{2} is the vector sum of the horizontal displacement d_{1} and vectorial displacement d_{2} . If $\frac{dd}{dt} = M(r)$ is used to evaluate initianal beaming measures (see the $d^{2} \neq \frac{d}{dt}$ in the case of large deflection). External beaming measures M(r) is not equal to P(L - r) but to P(L - r), where d_{1} is the sourcement of the horizontal discussion.

The modified form of Euler-Bernoulli equation is then given by

$$EI \frac{v^*}{[1+(v')^*]^{1/2}} = M(x) \qquad (5-5)$$

In the above, the curvature term is no longer $v' = d^2y/dx^2$. Instead, it became $v'/[1 + (v')^2]^{1/2}$. The bending moment M(x) on the other side of the equation should include the effect of changing length.

5.3 Scope of the Analysis

The beam analyzed in this shaper here deflections that maps (from 5% to nextly 55% of the beam length. It is ansmund that the concentrated linear applied on the summarized masses and the start of the sum of enteriors in a shared significantly due to trage deflection. It must be noted that the calculation is for "bald" deficient and not simply the vertical deflection. The following assumptions are applied bio first boundings.

- 1. The material is isotropic and homogeneous
- The material is relatively soft. Material failure or plasticity does not govern the deflections.
- 3. The calculation is for "total" deflection and not simply the vertical deflection.
- Shear effect is negligible. Only bending stresses and axial stresses govern the deflection. Axial stresses may develop if the cross-Section rotation is significant.
- 5. Cross-sections maintain their original shapes.
- 6. Lateral buckling is not considered.

The total deflection is a vector sum of vertical and horizontal deflections, an mentioned earlier. The horizontal deflections are influenced by change in length as well as the curvature of the beam. For simple cases, the change in length can be shown to be

$$\delta_{aust} = \int_{0}^{1} \frac{Pv'}{AE} dx$$

For a cantilever, the integral evaluates to

$$\delta_{mint} = \frac{Pr(x)}{AE} = \frac{P}{AE} \frac{PL^2}{6EI} \left[1 - 3\frac{x}{L} - \left(1 - \frac{x}{L}\right)^2 \right] \quad (5.7)$$

(5-6)

Similarly, we can find the horizontal displacement u(x) due to curvature of the beam using,

$$u_{commun} = \int \frac{dx}{\sqrt{1 + (y')^2}}$$
(5-8)

All these equations (e.g., eq. 5.2.5) can be solved using well known monetical methods meth an Range-Katta (Oder 4), eac, is combination with "shoring" equations and assumetable (Aduka) (2004), Application of such methods becomes complicated if the deflections are being compated for more elaborate cases such as famess. In the following, finite elament models have been used to run linear, including and perhap dependent non-definite and definition of heam mortextex.

5.4 The Direct Secant Technique

The direct secant technique presented earlier by Adluri [2001], Bolar and Adluri [2005], e.e., has been used in Chapter 3 for plantic limit load estimation. Similar idea are employed in Chapter 4 in a preliminary effort at estimating backling loads of finance.

Fig. 5.2 describes the basic idea of secant analysis. The initial linear analysis predicts stress and strain based on the original elastic modulus. If the stress exceeds

a set value (such as the yield limit), the material modulus is altered using a secant line. A reanalysis is carried out and the process in continued iteratively till convergence.

The secant modulus is updated iteratively as

$$E_{int} = \frac{\sigma_{\mu}}{\sigma_{\mu}}E_i. \quad (5-9)$$

The direct secant analysis of Adlari [2001] may not change the modulus. Instead, it may change any item that is directly proportional to secant stiffness such as moment of incrita, rigidity, or even the stiffness matrix itself [Laha and Adlari, 2005]. In that seem, secant stiffness for the acci tension can be expressed as

$$K_1 = \eta K_0$$
, (5-10)

where, K_i is the original stiffness, K_i is the new stiffness and η is the modification factor.

The analysis input files for ANSYS are given in the Appendices.

5.4.1 Cantilever Beam

A conflicter base molycie to a concentrate land at the first end is thrown in Fig. 31. The Bin barm in molycole for informe and nonlinear cases. The algorithm work in Chapter 4 have been employed for obtaining the robust sectar analysis results. The results are solvers in Fig. 3.5. The picture of a sector field for the maximum distribution of the sector operation of the sector of the sector information theory (the slope of the results of the lines theory on this graph should be 1:1). The robust analysis results are field y class to those from fill conflict analysis. Up to a result of the slope of the results of the lines theory on this graph should be 1:1). The robust analysis results are field y class to those from fill conflict analysis. Up to a result of the slope of the results of the lines theory on the scene starts to incruse when the deflection incruses beyond half the length of the beam. It is deemed that this is probably the most that we can expect in a normal structural application even for relatively "soft" materials. It is to be noted that the deflection includes the shortening of the lever arm as well as the tensile component of the applied load at anged functions.

Fig. 5.4 shows the comparison between the nonlinear results, linear results and a simplified algorithm which considers the secont modification η in Eq. 5.4.2 to be inplied procedure virtually gives the same results are done from datalied secont molysis. The good fit seconts to be mainly because the nonlinear results are relatively linear even at high databasements. Also, same-that anceptendly, the nonlinear different insolution of the field instance in the relatively finance even at high databasements. Also, same-that anceptendly, the nonlinear deflection is some from the non-theorement anceptendly, the nonlinear deflection is some from the non-theorement.

5.4.2 Simply Supported Beam with a Roller Support

A single speeded beam with a still sense support on one cell a lower in Fig. 5.5. A concentrated force is applied at the reld space of the beam. The roller on the simply supported bars, in its charge and perform the branching displacement and relation the versical movement. When large deflection occurs at the reld space, the test displacements is influenced by both the versical deflection and horizontal movement of the beam.

The (normalized) results are shown in Fig. 5.6. For this case, unlike the castilever case, the nonlinear deflection is larger than that from linear theory. The secant analysis approximates the nonlinear result fairly closely.

For this case also, a simplified approach was attempted similar to that used in the

case of cattlever. The factor q was faund in be 0.95. The results for this implified approach are virtually identical to those from secaret analysis. At this stage, it is unknown why the simplified approach works except for the obvious means that the difficulties, even in case of animator analysis, is almost from a diffitence a uniform reduction of stiffness (through the use of moment of inertio might somehow work. We need to researcher that the difficulties direction and are disclosed in the direction and are stiffness (through the use of moment of inertio might somehow work. We need to researcher that the difficulties direction and are the direction and are stiffness (through the uses) of moments of the direction and are stiffness (through the uses).

5.4.3 Simply Supported Beam without Roller Support

A simply supported beam without the roller support is shown in Fig. 5.7. The beam is subject to a concentrated force at the mid span. Because there is no roller at the support, the horizontal displacement is constricted. The concentrated force at the mid span only leads to the vertical defection.

Fig. 5.8 shows the (normalized) results of the analysis. As in the case of the previous beams, the secant analysis results are acceptable. For this case also the simplified technique was tried. A value of η =0.55 gives good results. Again, we are simply calibrating the value of η by using molinear analysis. We do not yet how the theoretical basis for the use of uniform constant molifordion.

A comparison of the results for simply supported beam with and without rollers is shown in Fig. 5.9. The difference is obviously not very much. However, this difference must be seen in light of the results for linear analysis shown in the same graph.

5.4.4 Portal Frame with Lateral Load

A simple portal frame with a lateral load at the beam level is shown in Fig. 510. It has the same height as the width. Both beam and column have the same properties. The large deflection analysis and secant analysis results are shown in Fig. 511. As can be seen, the secant analysis gives reasonably close result to that from the molanear analysis.



















Fig. 5.6 Behaviour of a Simply Supported Beam with One Roller



Fig. 5.7 Total Large Deflection of a Simply Supported Beam without Roller



Fig. 5.8 Simply Supported Beam without Roller



....







Chapter 6 Conclusions

6.1 Introduction

Limit load estimates are very useful for many engineering applications -both in design and analysis type problems. There has always been a need for robust methods for plastic limit load analysis from the point of view of numerical stability and effort. Robust limit load analysis has gained considerable attention over the past several years. Available robust methods adopt secant modulus modification as a means to cause redistribution in an elastic structure thereby producing limit behavior. The most significant among these methods are the r-node method, elastic compensation method and the ma family of methods. All of these use the von Mises vield criterion to define an effective stress. This effective stress is used to obtain an estimate of secont modulus. The r-node method involves identification of r-node peaks to obtain limit loads. Such identification might require considerable judgment in some cases. The elastic compensation method is based on a maximum stress value. Because of numerical local errors, it can sometimes be difficult to properly identify the failure mechanism and the consequent limit load. The m, family of methods have better theoretical basis but is more involved than the other methods and is quite conservative in many cases and unconservative in several other situations -especially for bending type problems. All of these modulus modification methods need stress level modifications and consequent discretization requirements that are very elaborate. The present thesis made use of a robust method which has several features of the above mentioned rebut techniques for the estimation of limit loads above with additional advances. The method generation for advances of the excising velocit methods so that is an the adjusted for any yield cluthonian and any finite element type (Jadani, 1992, 2001). It has previously been shown to work again well for beam times of attractor generation of forces such as moments and advances. The extension of non behavior of path-thell types. The generalizations uses sately adjust cluther that and a least an accurate selection that where the source of the star the source of a least an accurate selection that where the source. The next of this tachtings has been demonstrated in the spectre vestiges methods. It is easier (and charged to a performing selection that, sheafts beachting and the large differences.

The method has a good theoretical basis for plastic limit loads. However, the theory for buckling and large deflections needs considerably more work in order to be firmly established.

6.2 Summary

Chapter 2 of the thesis gives an over view of the limit theorems, buckling, large deflections, etc. It reviews the material on the current robust methods such as the node technique, m, technique, etc.

Chapter 3 reviews the methods used by Adluri and associates to estimate plastic limit loads for beam and plate type structures. The methods are used to predict frame collapse mechanisms and their limit loads. The basic procedure is taken from Adluri 11999. 20011 and is summarized below: 1. Carry out the initial elastic analysis based on original geometrical properties.

2. Modify the cross-section properties using

$$I_{new} = \frac{M_{\gamma}}{|M_{\gamma}(x)|}I_{dit}, \quad (6.1)$$

where M_{p} are the plastic moment capacities of the cross-sections. The bending moment $M_{q}(x)$ correspond to the elastic results from initial analysis.

- 3. A second elastic analysis is carried out based on this modification.
- 4. From the second analysis, the peak bending moments are considered as the locations of plantic hinges. Not all of these locations may be needed to form a mechanism for collapse. Sufficient contributions of these hinge locations are selected to form all possible hinge mechanism. Usually, there are only a forunda possible homings mechanism. Usually, there are only a forunda possible homings mechanism.
- For each of the selected mechanisms, the load factor for plastic collapse is calculated from the expression:

$$\lambda = \frac{P_{\perp}}{P} = \frac{1}{n} \frac{1}{\sum_{n=1}^{n} \frac{M_{post-ove}}{M_{p_n}}}$$
(6-2)

For uniform structures, M₁₀ is constant. Therefore the equation can be simplified

35:

$$\lambda = \frac{P_{\perp}}{P} = \frac{M_{P}}{M_{path-ph}}$$
(6-3)

This is the same as expression as that from Adluri [2001].

 The lowest of these load factors amongst all possible combinations for hinge mechanisms is the applicable plastic limit load factor.

The results of the above procedure have been compared against the full nonlinear analysis results for many types of frames including portal frames, one-storey frames with multiple bays, and multistory frames.

Chapter 4 applies two adapted versions of the above secant technique to predict the buckling loads of elastic frames.

- 1 it is amough that the frame backling does not follow the clinicial mode of "bifurcation" or a sudden change in status. Rather, the backling is a tunition from more are less tractistical distributions (newholing both statistical and lateral displacements of members) is a statisticity very large set of facual notineer displacements due to instability. The magnitude of this "large" displacement to or do a non-independence as the facta that the short and. This is similar to the fact that as plancic limit bands, the curvature of the beam set critical arguments reacher very large values (in magnitude is not important -ather that it took plane at ath).
- 2 The relationship between applied bash and the displacement at critical location in somewhat analogous to that between moment and curvature of the critical cross-section in Chapter 3. Both the relationships have initially linearly increasing portions which turn to large horizont portions (with little increase in effective load).
- 3 Just as we obtain locations of peak redistributed moments in Chapter 3, we obtain locations of neak displacements in the structure whose stiffness is modified by the

secant scheme. These peak displacements can potentially be indicative of frame instability.

4 Two alternative empirical methods have been used in this chapter. Method 1 computes peak displacements as given below:

$$d_{apc} = C \frac{M_{abp-post}}{P_{commund}}$$
(6-4)

The load factor λ for the critical load P_{ar} is obtained from the peak displacements,

$$\mathbf{P}_{cr} = \mathbf{P} \lambda_{u} \lambda = \frac{1}{n} \frac{\sum_{i=1}^{n} d_{sys}}{\sum_{i=1}^{n} d_{sust}}$$
(6-5)

where, d_{and} is the peak displacement in the reanalysis.

The factor C depends on the type of frame. We do not yet have the necessary theoretical development to determine this factor exactly. Pending the development of the theory, it has been decided to empirically estimate the factor using finite element econettics confinement and/sis.

In empirical method 2, after reanalysis, the critical load is calculated as below:

1 Identify the locations of peak displacements and the nearby locations of zero displacements. Find the bending moments and displacements at these locations.

2 Calculate

$$\Delta M = M_{der} - M_{de}$$
 and $\Delta d = d_{er} - d_{de}$

where, d_{σ} is the peak displacement in reanalysis, d_{σ} is the zero displacement, $M_{d\sigma}$ is bending moment at the peak displacement location and M_{\pm} is the bending moment at zero displacement location.

3 Estimate the critical load by using the equation

$$P_{cr} = \beta \frac{\Delta M}{\Delta d}$$

(6-7)

The factor β is analogous to the factor C in Method 1. It is calibrated using the same nonlinear analysis as used in Method 1.

The methods have been used to estimate the backling loads for several different portal frames. Further work is needed to establish the theoretical basis for these methods and to agely them (or improved versions of them) for more complicated frames.

Chapter 3 datas with large addression of beams and firmers. The same techniquers as in Chapter 3 are used to predict the large deflections of beams up to a value of nearly 50% of the apass houghts. The results are encouraging. A molification form of the scenar technique where the modification is uniform throughout the length also scenar to give quite acceptable results. However, as in the case of Chapter 4, nove work is results to acceled to scatishis the theoretical basis for the methods.

The analyses in the thesis are carried out using ANSYS software. Typical input files for all different types of analyses and the APDL routines required are provided in the Amendices.

6.3 Conclusions

There were several studies in the current thesis. Some of the main conclusions are

6.3.1 Plastic Limit Loads Estimation

- The best conclusion of the plantic collapse study of the frames in Chapter 3 is that the method works very nicely and had a sound theoretical basis. The errors compared to the full nonlinear analyses are well within acceptable margins.
- 2. The melods works equally well for fumes with non-unitered memory of the fumes, single storage multi-hose fumes, and multimerry frames. They include beam mechanism failures and a ways methadiom failures are well as combined failure mechanisms. The studies very clearly confirm what has been suggested or claimed by cartler studies. This is perhaps the best contribution of the present thesis.

6.3.2 Buckling

- The load-displacement relationship in a buckling problem is analogous to the relationship between moment and curvature in a plastic limit load estimation problem.
- We can modify moment of inertia (in lieu of the stiffness) based on the lateral displacement profiles in order to simulate critical displacement patterns at backling.
- 3. Two empirical methods have been examined in Chapter 4 to predict the

backling load capacities of portal frames. Factors C and β have been calibrated for use in Eq. 4.22 and eq. 4.45. These factors are indicative of certain theoretical algorithms for backling loads. The results using these seem to be fairly good for a range of portal frames and loadings. However, their theoretical basis needs further study.

6.3.3 Large Deflection Analysis

- For beam like structures studied in this thesis, the linear analysis deflections could be either larger or smaller than those predicted by nonlinear analysis. For cantilever, nonlinear analysis gives less deflection than the linear analysis. For other beams and frames, it is over the other way around.
- For the examples studied, linear and nonlinear analyses do not differ much till the deflection reaches about 20% of the span. Even after that, the difference is not very large till the deflection exceeds 30% of the span.
- 3. The direct secant approach gives reasonably good estimate of the nonlinear deflections. However, many more examples need to be studied to confirm this observation. This is because it can be argued that even linear analysis is fairly close to the results of nonlinear analysis.
- 4. A uniform secant modification for the entire beam (instead of deflection based modification that gives a non-uniform beam cons-section) scents to give results that are almost identical to the full secant analysis. The uniform modification factor used for moment of inertia has a narrow range: 0.95 < r < 1.05. If linear deflection is guerer than total nonlinear deflection.</p>

143

 η > 1.0. If linear deflection is smaller than total nonlinear deflection, η < 1.0. This needs to be studied further and theoretical basis established in order for it to be of practical use.

6.4 Recommendations for Further Work

In the present thesis, the research involved limit load entimistics due to plastic collapse, excitational load entimation due to classic backling and entimation of classic load effections in beam. The weak is limited to central type of structures. More needs to be carried out to condulate the validity of these approaches. Chapter 3 is portupe with the more theoretical justifications as given by Adlant [2001]. Chapter 4 and 5 need considerable further research to cendificat their further transmission with its resonanded to the following context.

- 1. Improve the automatic identification of plastic hinge mechanisms.
- Extend the robust secant estimation of Chapter 4 to include material nonlinear analysis.
- 3. Extend the robust secant estimation of buckling to more complicated structures.
- Factors C and β are used to calculate the critical load. The meaning of the factors needs to be firmly established and made available for different situations.
- Large deflection estimation needs to be extended to more complicated structures with different boundary conditions and load combinations.

REFERENCES

Adluri, S.M.R., 1999. Robust Estimate of Limit Loads of Steel Structures Using Modified Geometric Properties of Members, MUN EAST Report No: 99-1, Eng & App Sc Res. Rep. Series, Fac. of Engrg, Memorial University, St John's, Canada.

Adluri, S.M.R. 2001a. Basic Secant Technique for Limit Loads and Nonlinear Analyses. Res. Rep.02-01, Fac. of Engrg, Memorial University, St John's, Canada.

Adluri, S.M.R. 2001b "Direct Estimation of Limit loads Using Secart Rigidity and Scald Yield Criteria," CANCAM 2001, Canadian Conf. App. Mech., St. John's,

Adluri, S.M.R. 2007. Finite Element Analysis for Engineering Applications, ENG 9501 Geaduate Course Notes, Faculty of Engineering, Memorial University, St. John's, Canada.

Adluri, S.M.R. 2009. "Direct Estimation of Limit loads Using Secant Rigidity and Scald Yield Criteria," (Under View).

Adluri, S.M.R. 2009. "Applied Mathematical Analysis," ENG 5434 class notes, Faculty of Engineering, Memorial University, St. John's, Canada.

AISC, 2006. AISC- ASD/LRFD Seeel Construction Manual w/CD-ROM, 13th Edition.

Allen, H.G., 1980. Background to Backling, McGraw-Hill Book Company (UK) Ltd.

ANSYS, 2005. Finite element software, ANSYS Inc., PA, USA.

Bathe, K.J., 1996. Finite Element Procedures, Prentice-Hall, Inc., Englewood Cliffs, New Jersey, USA.

Barnes, J.G.P., 1965. "An Algorithm for Solving Nonlinear Equations Based on the Secant Method," Computer & Structures, Vol. 17, No. 5-6, pp. 871-879. Bolar, A.A., 2001. Robust Estimation of Limit Loads of Plates Using Secant Rigidity., M.Eng. Thesis, Faculty of Engineering, Memorial University of Newfoundland, St John's, Newfoundland, Canada.

Bolar, A.A., and Adluri, S.M.R., 2001. "Limit Load Estimation for Plates Using Secant Rigidity," CANCAM 2001. Canadian Conf. APP. Mech., St John's. NF.

Bolar, A.A., and Adluri, S.M.R., 2005. "Limit Load Estimation for Plate Using Secant Rigidity," CSCE Annual Conf. Toronto.

Boyle, J.T., Hamilton, R., Shi, J., and Mackenzie, D., 1997. "A Simple Method of Calculating Lower Bound Limit Loads for Axisymmetric Thin Shells," *Transactions* of the ASME, Vol. 199p;e236-242.

Chajes, A. 1993. Principles of Structural Stability Theory, Waveland Pr Inc ISBN-10: 0881337382

Chen, W-F., and Lui, E.M., 1987, Structural Stability Theory and Implementation, Elsevier Science Publishing Co., Inc., New York.

Chen W-F., and Lui, E.M., 1991. Stability Design of Steel Frames. CRC Press, Inc., Boca Raton, Florida.

Chen, W-F, and Atsuto, T., 1976. Theory of Beam-Columns, McGraw-Hill Inc., USA.

CISC, 2008. Eight Ed. Hand book of Steel Construction. Canadian Institute of Steel Construction, Willodale, Ontario, Canada.

Dhalla, A.K., and Jones, G.L. 1986. "ASME code classification of pipe stresses: A simplified elastic procedure," *International Journal of Pressure Vessels and Piping*, Vol. 26 pp. 145-166.

Dhalta, A.K., and Jones, G.L. 1984. "Verification of an elastic procedure to estimate elastic follow-up," *Design of Elevated Temperature Piping, Pressure Vessel and piping Division, ASME, New York*, Vol. 86pp. 81–96. Edmand, R.F., 2000. Robust Methods of Finite Element Analysis: Evaluation of Non-Linux Lower Bound Limit Loads of Plated Structures and Stiffening Members. M.Eng. thesis, Faculty of Engineering Memorial University of Newfoundland, St John's Newfoundland, Canada.

Fernando, C.P.D, 1992. Limit Loads of Mechanical Component and Structures Using The Gloss R-node Method, M.Eng. thesis, University of Regina, Saskatchewan, Canada.

Fowler, C.G., 1998. Robust Extimation of Limit Loads for Cracked Components. M.Eng. thesis, Faculty of Engineering , Memorial University of Newfoundland, St John's, Newfoundland, Canada.

Hibbeler, R.C., 2008. Mechanics of Materials. Seventh Ed. Person Prentice Hall Person Education, Inc. Upper Saddle River, New Jersey 07458.

Hunt, G.W. 1984. Elastic Instability Phenomena., John Wiley and Sons, Inc., Chichester.

Iyengar, N.G. R., 1988. Structural Stability of Columns and Plates, John Wiley and Sons.Inc., Chichester.

Kizhatil, R.K., 1993. Robust Approximate Methods for Inelastic Analysis of Mechanical Components, Ph.D. Dissertation, University of Regina, Canada.

Kizhatil, R.K., and Seshadri, R., 1991. "Inelastic Strain Concentration Factors and Low Cycle Fatigue of Pressure Components Using Gloss Analysis," *Proceeding of the* ASME-PVP Conference. San Diego, Vol. 2010, pp 149-154.

Laha, D.C., and Adluri, S.M.R., 2005. "Inelastic Strain Estimation Using Direct Secant Method." CSCE Annual Conf. Toronto.

Li, P. 2003. Limit load estimation for structures under mechanical loads. Ph.D. Dissertation, Faculty of Engineering, Memorial University, St John's, Canada. Lin, T.H., 1968. Theory of Inelastic Structures, John Wiley and Sons ,Inc., New York.

Logan, D.L., 2007. A First Course in the Finite Element Method, Fourth Ed., Thomson Pub., ISBN 0-534-55298-6

Mackenzie, D., and Boyle, J.T., 1993. "A method of estimating limit loads by iterative elastic analysis. I-simple examples," *Journal of Pressure Vessel Technology*, Vol. 53 pp. 77-95.

Mackenzie, D., and Boyle, J.T., and Hamilton, R., 2000. "The elastic compensation method for limit and shakedown analysis: a review," *Journal of Strain Analysis*. Vol.35 No. 3, pp. 171-188.

Mackenzie, D., and Nadurajah, C., shi, J., Boyle, J.T., 1993. "Simple Bounds on Limit Loads by elastic Finite Element Analysis," *Journal of Pressure Vessel Technology*, Vol. 115 pp. 27-31.

Marriott, D.L., 1988. "Evaluation of Deformation or Load Control of Stress under Inelastic Finite Element Stress Analysis," *Proceeding of ASME-PVP Conference*. Pittsburge, Vol.136.

Mangalaramanan, S.P., 1997. Robust Limit Loads Using Elastic Modulus Adjustment Techniques. Ph.D. Dissertation, Faculty of Engineering, Memorial University, St. John's, Canada.

Moaveni, S. 2007. Finite Element Analysis Theory and Application with ANSYS. Third Ed. Prentice Hall, ISBN-13: 9780131890800.

Mura, T., and Lee, S.L., 1962. "Application of variational Principle to Limit Analysis," *Quarterely of Applied Mathematics*, Vol.21, No. 3, pp.243-248.

Mura, T., Rimawi, W.H., Lee, S.L., 1964. "Extended Theorems of Limit Analysis," *Owarterely of Applied Mathematics*, Vol.23, No.2, pp.171-179.

Neal, B.G., 1977. The Plastic Methods of Structure Analysis, Third Ed, Chapman

and Hall Ltd., New York,

Nadanjah, C., Mackenzie, D., and Boyle, J.T., 1993. "A Method of Estimating Limit Loads by iterative elastic Analysis. II- nozzle sphere intersection with internal pressure and radial load," *International Journal of Pressare Vessels and Piping*, Vol. 35 no.97-119.

Neal B.G., 1977. The Plastic Methods of Structural Analysis, Third Ed, Chapman and Hall Ltd., New York: Amsterdam: London.

Owen, D.R.J., and Gomez, C.M.B., 1981. "An Approach of Numerical Solution Techniques: for Elasta-Plastic and Elasta-Viscoplastic Material Problems," *International Conference on Structures Mechanism in Reactor Technology*, Paris, France.

Pan, Li., 2003. Limit load estimation for structures under mechanical loads. Ph.D. Dissertation, Faculty of Engineering, Memorial University, St John's, Canada.

Powell, G., and Simons, J., 1981. "Improved Iteration Strategy for Nonlinear Structures," *International Journal for Numerial Methods in Engineering*, Vol.17. pp. 1455-1467

Roberts, T.M., and Ashwell, D.G. 1971. "The Use of Finite Element Mid-Increment Stiffness Matrices in the Post-Buckling Analysis of Imperfect Structures." *International Journal Solid & Structures*, Vol. 7, pp. 805-823.

Ralph, F.E., 2000. Robust Methods of Finite Element Analysis: Evaluation of Nonlinear, Lower Boundary Limit Loads of Plated Structures and Stiffness Members. M.Eng. Thesis, Faculty of Engineering, Memorial University, St John's, Canada

Roorda, J., 1980. Buckling Elastic Structures. Solid Mechanics Division, University of Waterloo Press.

Seshadri, R., 1991. "The Generalized Local Stress Strain (GLOSS) Analysis- Theory and Applications," *Journal of Pressure Vessel Technology*, Vol. 113 pp. 219-227. Seshadri, R., 1997. "In Search of Redistribution Nodes," International Journal of Pressure Vessels and Piping, Vol. 73 pp. 69-67.

Seshadri, R., and Bubu, S., 2000, "Extended GLOSS Method for Determining Inelastic Effects in Mechanical Components and Structures: Isotropic Material," *Journal of Pressure Vessel Technology*, Vol. 122, pp. 413-420.

Seshadri, R., and Fernando, C.P.D., 1992. "Limit Loads of Mechanical Components and Structures Using the Gloss n-node Method," ASME Journal of Pressure Vessel Technology. Vol. 114, pp. 201-208.

Seshadri, R., and Mangalazamanan, S.P., 1997. "Lower bound limit loads using variational concepts: the m₈ method," *International Journal of Pressure Vessels and Plotog*, Vol. 71, pp. 93-106.

Seshadri, R., and Marriott, D.L., 1992. "On relating to the Reference Stress. Limit Load and the ASME stress Classification Concepts," *Journal of Pressure and Piping*, Vol. 230, pp. 135-149.

Seshardri, R., and Wu, S., 2001, "Robust Estimation of Inelastic Fracture Energy Release Rate (J): A Design Approach," *Journal of Pressure Vessel Technology*, Vol.123, pp.214-219

Shames, I.H., Cozzarelli, F.A., 1992. Elastic and Inelastic Stress Analysis, Prentice-Hall International. Inc., Englewood Cliffs, NJ

Stricklin, J.A., and Haisker, W.E., 1977. "Formulations and Solution Procedures for Nonlinear Structural Analysis," Computers & Structures, Vol. 7. pp.125-136.

Thomas, G.R., 1973. " A Variable Step Incremental Procedure," International Journal for Numerical Methods in Engineering, Vol. 7. pp. 563-566.

Timoshenko, S.P. and Woinowsky-Krieger. E. 1975. Theory of Stability, McGraw-Hill Co., Trahair, N.S., 1993. Flexural Torsional Backling of Structures.CRC Press Inc.,2000 Corporate Boulevard NW, Boca Raton, Florida 33431,USA.

Ugural, A. C., and Fenster, S.K., 1987. Advanced Strength and Applied Elasticity, Elsevier Pub. ISBN-10-0444010661

Vallianppan, S., 1972, "Elasto-Plastic Analysis of Anisotropic Work-hardening Material," Archives of Mechanics, Vol. 24, No. 3, pp. 465–481.

Wang, C.M., Reddy, C.Y., and Reddy, J.N., 2005. Exact Solutions for Buckling of Structural Members, CRC Press, Boca Raton, Florida.

Wellford, L.C., and Sen, H., 1981. "A Selective Relaxation Iterative Solution Technique for Nonlinear Structural Analysis Problem," *International Journal for Numerical Methods in Engineering*, Vol. 17, pp. 773-795.

Wolfe, P., 1959. "The Secant Method for Simultaneous Nonlinear Equations," Communications of the ACM, Vol. 2, pp. 12-13.

Xue, L.P., 2006, Elastic and Inelastic Finite Element Analysis of Cylindrical Shell Intersections. Ph.D. Dissertation, Marquette University, USA.

Zhang, Y.G., and Lu, M.W., 1995, "An Algorithm for Plastic Limit Analysis," Computer Method in Applied Mechanics and Engineering, Vol. 126, pp.333-341.

APPENDICES

Appendix A includes all the first purely datatic analyses in nother scenar methods in this research. The macross date for modeling scenar rightly are all presented in Appendix II. Appendix and appendix II are used toughter to constant of the entire robust scenar analysis. Appendix C involves all the FLA meditmer analyses in this research. The results of robust scenar multiysis obtained from Appendix A and B are compared with the entire scenar from Appendix C. C

APPENDIX A

A.1.1 A UNIFORM PORTAL FRAME SUBJECT TO A LATERAL

FORCE AND A VERTICAL FORCE

PLASTIC ANALYSIS USING MODIFIED SECANT RIGIDITY PREP7 : ENTER PREPORCESSOR

ET, 1,beam3 *SET,EM, 200E3 *SET,A, 50 *SET, 1, 266.67 *SET, H, 8 *SET,LZ,10 *SET,P1, 8e2 *SET,P2, 7,2e2

R.I.A.I.H

MP,EX, 1, 200e3 MP, NUXY,1.0

K,1, 0, 0 K,2,0,400 K,3,400,400 K,4,400,0

L,1,2 :DEFINITION OF LINES L,2,3 L,3,4

LESIZE, ALL, LZ LMESH, ALL

NSEL,S,LOC,X,200 F,ALL,FY,-P1

NSEL,S,LOC,X,0 NSEL,R,LOC,Y,400 F,ALL,FX,P2 :USE BEAM 3 ELEMENT :YOUNG'S MODULUS :AREA : MOMENT OF INERTIA : HEIGHT : MESH SIZE : VERTICAL FORCE : LATERAL FORCE

INPUT AREA, MOMENT OF INERTIA AND HEIGHT INPUT YOUNG'S MODULUS INPUT PASSION'S RATION

IDEFINITION OF EYPOINTS

DEFINITON OF MESH SIZE MESH LINES

APPLY THE VERTICAL FORCE

APPLY THE LATERAL FORCE

APPLY BOUNDARY CONDITION

NSEL,SLOC₃,0 D,ALL,ALL NSEL,ALL /SOLU ANTYPE 0

SAVE SOLVE ! DEFINE STATIC ANALYSIS

A.1.2 A TWO BAY AND ONE STOREY FRAME SUBJECT TO

TWO VERTICAL FORCES AND ONE LATERAL FORCE

PLASTIC ANALYSIS USING MODIFIED SECANT RIGIDITY

/PREP7 ET. 1. beam3

*SET, EM ,200E3 *SET, A, 50 *SET, I, 266.67 *SET, H, 8 *SET,LZ,10 *SET,P1, e2 *SET,P2,7.2e2 *SET,P3, 5-2

R,1,A,I,H MP,EX,1,200e3 MP,NUXY, 1, 0 : ENTER PREPORCESSOR ! USE BEAM 3 ELEMENT ! YOUNG'S MODULUS ! AREA ! MOMENT OF INERTIA ! HEIGHT

MESH SIZE VERTICAL FORCE VERTICAL FORCE

1 INPUT AREA, MOMENT INITIAL 1 INPUT YOUNG'S MODULUS 1 INPUT PASSION'S RATION

DEFINITION OF KEYPOINTS

K,1,0,0 K,2,0,400 K,3,400,400 K,5,500,400 K,6,500,0 L,1,2 L,2,3 L,3,4 L,3,5 L,5,6 LESIZE, ALL, 10

LESIZE, ALL, 10 LMESH, ALL

NSEL, S. LOC, X. 200

: DEFINITION OF LINES

1 DEFINITON OF MESH SIZE 1 MESH LINES

1 APPLY THE VERTICAL FORCE

154

F. ALL, FY.-P1

NSEL,S,LOC,X,600 F,ALL,FY,-P2 APPLY THE VERTICAL FORCE

1 APPLY THE LATERAL FORCE

NSEL,S,LOC,X,800 NSEL,R,LOC,Y,400 F,ALL,Fx,P3

APPLY BOUNDARY CONDITIONS

NSEL,S,LOC,y,0 D,ALL,ALL NSEL,ALL

/SOLU ANTYPE,0 :DEFINE : SOLVE FINISH /NPUT MACRO1

A.I.3 A UNIFORM FRAME SUBJECT TO THREE VERTICAL FORCES AND ONE LATERAL FORCE

PLASTIC ANALYSIS USING MODIFIED SECANT RIGIDITY

PREP7 ET, LBEAM3 ENTER PREPORCESSOR

*SET, EM, 20083 *SET, A, 50 *SET, I, 266,67 *SET, II, 8 *SET, II, 8 *SET, II, 8 *SET, II, 8 *SET, 22, 25 *SET, 12, 20083 MP, NUXY, 1, 0 K, 10,0 K, 20,0 K, 20,0 K, 4,0,3 *L,0

K,5,0,2*L K,6,0.8*L,2*L YOUNG'S MODULUS AREA MOMENT OF INERTIA HEIGHT LENGTH MESH SIZE FORCES

INPUT AREA, MOMENT INITIAL

1 INPUT PASSION'S RATION

DEFINITION OF KEYPOINTS
K,7,1.3*L,L K,8,1.3*L,0	
L12	2 DEFINITION OF LINES
1.2.3	
1.3.4	
1.2.5	
L.5,6	
1.6.3	
1.3.7	
L,7,8	

LESIZE,ALL,LZ LMESH,ALL /SOLU DEFINITON OF MESH SIZE

NSEL,S,LOC,X,0.4*L NSEL,R,LOC,Y,2*L EALL-FY-P1

NSEL,S,LOC,X,1.05*L NSEL,R,LOC,Y,L F,ALL,FY,-P2 1 APPLY THE VERTICAL FORCE

APPLY THE VERTICAL FORCE

APPLY THE LATERAL FORCE

NSEL,S,LOC,X, 0 NSEL,R,LOC,Y,2*L FALL,Fx,P2

NSEL ALL

NSEL,S,LOC,y,0 D,ALL,ALL NSEL,ALL APPLY BOUNDARY CONDITIONS

/SOLU ANTYPE,0 SOLVE FINISH /INPUT MACRO1

DEFINE STATIC ANALSIS

A.1.4 A UNIFORM BIG FRAME SUBJECT TO UNIFORMLY LATERAL DISTRIBUTED FORCES AND VERTICAL CONCENTRATED FORCES

PLASTIC ANALYSIS USING MODIFIED SECANT RIGIDITY PREP7 STREPREPORCESSOR

PREP7 ET, 1,BEAM3 LENTER PREPORCESSOR USE BEAM 3 ELEMENT

*SET, EM ,200E3 *SET, A, 50 *SET, J,266.67 *SET, H,8 *SET,L,150 *SET,L2, 2 *SET,P1, 1c2 *SET,P2, 1.5c2 *SET,P2, 1.5c2 YOUNG'S MODULUS AREA MOMENT OF INERTIA HEIGHT LENGTH MESH SIZE FORCES FORCES

1 INPUT AREA MOMENT INITIAL

! INPUT PASSION'S RATION

DEFINITION OF KEYPOINTS

R,1,A,J,H MP,EX,1,200e3 MP,NUXY,1,0 K,1,0,0 K,2,1,2*L,0 K,3,2,4*L,0 K,4,3,4*L,0 K,5,4,4*L,0 K,5,4,4*L,0

K,6,0,L K,7,1.2*L,L K,8,2.4*L,L K,9,3.4*L,L K,10,4.4*L,L

K,11,0,2*L K,12,1.2*L,2*L K,13,2.4*L,2*L K,14,3.4*L,2*L K,15,4.4*L,2*L

K,16,1.2*L,3*L K,17,2.4*L,3*L K,18,3.4*L,3*L

K,19,1.2*L,4*L K,20,2.4*L,4*L K213.4*L.4*L

L,1,6	DEFINITION OF LINES
L,2,7	
L,3,8	
L,4,9	
L,5,10	
L,6,7	
L,7,8	
L.8.9	
L,9,10	
L.6.11	
1,7,12	
L.8,13	
L.9.14	
L.10.15	
L.11.12	
L,12,13	
L,13,14	
L,14,15	
1.12.16	
L.13.17	
L.14,18	
L-16.17	
L,17,18	
L-16-19	
1.17.20	
L-18.21	
1-19.20	
1.20.21	

LESIZE, ALL,LZ LMESH, ALL /SOLU ! DEFINITON OF MESH SIZE ! MESH LINES

ESEL,S,ELEM,,1,75 FORCE SFBEAM,ALL,LPRES,P1

ESEL,S,ELEM,,706,780 SFBEAM,ALL,1,PRES,P1

ESEL,S,ELEM,,1411,1485 SFBEAM,ALL,1,PRES,P1 : APPLY THE LATERAL

ESEL,S,ELEM, 1801,1875 SFBEAM, ALL, I, PRES, P1

ESEL,S,ELEM_1336,1410 SFBEAM,ALL,1,PRES,P2

ESEL,ALL,ALL

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,L F,ALL,FY,-P3

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,2*L FALL,FY,-P3

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,3*L F,ALL,FY,-P3

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,4*L F.ALL,FY,-P3

SEL,S,LOC,9,0
D,ALL,ALL
NSEL,ALL
ANTYPE,0
OUTRES, ALL, ALL
SAVE
NSEL.ALL

NSEL,S,LOC,y,0 D,ALL,ALL NSEL,ALL APPLY THE VERTICAL FORCE

APPLY BOUNDARY CONDITIONS

/SOLU ANTYPE,0 SOLVE FINISH /NPUT MACRO1

DEFINE STATIC ANALYSIS

A.1.5 A NON-UNIFORM PORTAL FRAME SUBJECT TO A

LATERAL FORCE AND A VERTICAL FORCE

PLASTIC ANALYSIS USING MODIFIED SECANT RIGIDITY

PREP7 ET. LBEAM3

*SET EM 200E3 *SET.L.800

*SET.A1.3000

*SET 11 9.4 *SET.H1.60

*SET.A2, 5000 *SET 12, 4.17e6

*SET,A3,7000 *SET.I3.1.143e7 *SET.H3.140

*SETLZ.5 *SET,P1,1E6 *SET,P2,1E6

K.1.0.0 K.2.0.1 K 3 3/24L L K.4.3/2*L.0

1.3.4 LSELS_1

LATT_2 LINE 1 LSELS_2

LATT-1

LINE 2

LINE 3

LSELS_3 LATT_3

1 ENTER PREPORCESSOR USE BEAM 3 ELEMENT

'VOLNG'S MODULUS

DEFINE DIFFERENT : GEOMETRICAL PROPERTIES

INPUT AREA MOMENT INITIAL

1 MESH SIZE

R.I.AI.JI.HI R.2. A2.12.H2 R 3 A 3 13 H3 INPUT YOUNG'S MODULUS UIMP.I. EX., EM MPNUXY1.0 'INPUT PASSION'S RATION DEFINITION OF LINES INFLECT LINE 1 APPLY REALCONSTANT 2 TO SELECTLINE 2 **APPLY REALCONSTANT 1 TO** SELECT LINE 3 **APPLY REALCONSTANT 3 TO** DEFINITION OF MESH SIZE LESIZE, ALL LZ

I MESH ALL /SOLU NSELS LOC X 0 NSEL.R.LOC.Y.L FALL-FX.P1 NSELSLOCX3/4*L MESH LINES

1 APPLY THE LATERAL FORCE

APPLY THE VERTICAL FORCE

NSEL R LOC VI. FALL EV.P2 NSELS LOC v 0

IAPPLY THE PROLINDARY CONDITIONS

DALL-ALL NSEL ALL ANTYPE.0 SOLVE /INPUT MACRO1

1 DEFINE STATIC ANALYSIS

A.2.1 A UNIFORM PORTAL FRAME SUBJECT TO THREE

VERTICAL FORCES ON THE BEAM

1 BUCKLING ANALYSIS USING MODIFIED SECANT RIGIDITY /PDEP7 ET.1.REAM3 JUSE BEAM 3 LEMENT

*SET.EM.200E3 *SET.L1.600 *SET1 2 800 4SET.A. 50 *SET 1416.67 DEFINITION *SET.H. 10

ASETIZ 5

WOUNC'S MODULUS **!LENGTH DEFINITION**

AREA DEFINITION MOMENT OF INERTIA

HEIGHT DEFINITION IMESH SIZE DEFINITION

*SET.P1.2.0E4 4SET P2 1 SE4 **SET P2 1 2E4** RIALH MPEX 1 200+3 TB. BKIN TRDATA, 1, 300.0

1 INPUT AREA, MOMENT INITIAL 'INPUT VOUNG'S MODULUS IDEEDNE BII INFAR MATERIAL DEFINE VIELD STRESS AND THE SLOPE AFTER VIEDL STRESS THE SLOPE AFTER VIEDL STRESS ! DEFINITION OF KEYPOINTS

k.1.0.0 k2011

k,3,L2,L1 K,4,L2,0

L,1,2 L,2,3 L,3,4 DEFINITION OF LINES

3MESH LINES

LESIZE,ALL,LZ LMESH,ALL /SOLU NSELS J.OC,Y.0

APPLY THE BOUNDARY CONDITION

D.ALL.ALL

NSEL,S,LOC, X, 0.3 *L2 NSEL,R,LOC, Y,L1 F,ALL,FY,-P1 1 APPLY THE VERTICAL FORCES

NSEL,S,LOC,X,0.5*L2 NSEL,R,LOC,Y,L1 F,ALL,FY,-P2

NSEL,S,LOC,X,0.8 *L2 NSEL,R,LOC,Y,L1 FALL,FY,-P3

NSEL, ALL ANTYPE,0 SOLVE FINISH /NPUT MACRO2

: DEFINE STATIC ANALYSIS

A.2.2 A UNIFORM PORTAL FRAME SUBJECT TO A VERTICAL

FORCE AND A LATERAL FORCE

BUCKLING ANALYSIS USING MODIFIED SECANT RIGIDITY /PREP7 ET.LREAM3 USE BEAM 3 LEMENT

*SET,EM,200E3 *SET,L1, 600 *SET,L2, 800

*SET, A, 80 *SET, I, 666.67 DEFINITION USE BEAM 5 LEMENT

YOUNG'S MODULUS LENGTH DEFINITION

AREA DEFINITION MOMENT OF IN

INERTIA

*SET,H, 10 *SET,LZ, 5 *SET,P1,2.0E4 *SET,P2,0.5E4

R,1,A,I,H MP,EX,1, 200e3 TB, BKIN TBDATA, 1, 300.0

k,1,0,0 k,2,0,L1 k,3,L2,L1 K,4,L2,0

L,1,2 L,2,3 L,3,4

LESIZE,ALL,LZ LMESH,ALL /SOLU

NSELS LOC.Y.0

D.ALL.ALL

NSEL,S,LOC, X, 0.5 *L2 NSEL,R,LOC, Y,L1 F,ALL,FY,-P1

NSEL,S,LOC,X,0 NSEL,R,LOC,Y,L1 FALL,FX, P2 HEIGHT DEFINITION MESH SIZE DEFINITION FORCE DEFINITION

INPUT AREA, MOMENT INITIAL INPUT YOUNG'S MODULUS DEFINE BILINEAR MATERIAL DEFINE VIELD STRESS AND THE SLOPE AFTER YIEDL STRESS THE SLOPE AFTER YIEDL STRESS DEFINITION OF KEYPOINTS

DEFINITION OF LINES

: MESH LINES

APPLY THE BOUNDARY CONDITIONS

APPLY THE VERTICAL FORCE

APPLY THE LATERAL FORCE

NSEL, ALL ANTYPE,0 SOLVE FINISH ANPUT MACRO2

1 DEFINE STATIC ANALYSIS

A.3.1 A CANTILEVER BEAM SUBJECT TO A CONCENTRATED

FORCE AT THE FREE END

! LARGE DEFLECTION ANALYSIS USING MODIFIED ROBUST SECANT METHOD

/PREP7 *SET_EM_200E3

*SET,A1, 0.01 *SET,I1,8.33e-6 *SET,H1,0.1 *SET,L1,1 *SET,L2, 1/50 *SET,P,(32)*1e2

ET,1,BEAM3 R,1,A1,I1,H1 MP,EX,1,200e3 MP,NUXY,1,0

K,1,0,0 K,2,L1,0 L,1,2 LESIZE,ALL, LZ LMESH,ALL

NSEL,S,LOC,X,0 NSEL,R,LOC,Y,0 D,ALL,ALL

!USE BEAM 3 LEMENT

AREA MOMEMNT OF INITIAL HEIGH LENGTH MESH SIZE FORCE

USE BEAM 3 ELEEMNT INPUT GEOMETRICAL PROPERTIES INPUT YOUNG'S MODULUS INPUT PASSION'S RATION

! DEFINITION OF KEYPOINTS

DEFINITON OF LINES DEFINITION OF MESH SIZE MESH ALL THE LINES

APPLY THE BOUNDARY CONDITIONS

APPLY THE VERTICALLOAD

NSEL,S,LOC,X, L1 NSEL,R,LOC,Y,0 F,ALL,FY,-P

NSEL, ALL ANTYPE, 0 SAVE /SOLU SOLVE FINISH ANPLIT MACROS

! DEFINE STATIC ANALYSIS

A.3.2 A SIMPLY SUPPORTED BEAM WITH ONE ROLLER SUPPORT SUBJECT TO A CONCENTRATED FORCE IN THE MIDDLE

! LARGE DEFLECTION ANALYSIS USING MODIFIED ROBUST SECANT METHOD

PREP7

*SET,EM,200E3

*SET,A1, 0.01 *SET,11, 8.33e-6 *SET,H1,0.1 *SET,L1,1 *SET,LZ, 1/50 *SET,P,(32)*1e2

ET,1BEAM3 R,1,A1,11,H1 MP,EX,1,200e3 MP.NUXY,1,0

K,1,9,0 K,2,L1,0 L,1,2 LESIZE,ALL, LZ LMESH,ALL

NSEL,S,LOC,X,0 NSEL,R,LOC,Y,0 D,ALL,UX D.ALL,UY

NSEL,S,LOC,X,L1 NSEL,R,LOC,Y,0 D,ALL,UY **!USE BEAM 3 LEMENT**

AREA IN MM² MOMEMNT OF INITIAL IN MM⁴ HEIGH IN MM LENGTH IN MM MESH SIZE IN MM FORCE IN NEWTON

USE BEAM 3 ELEEMNT INPUT GEOMETRICAL PROPERTIES INPUT YOUNG'S MODULUS INPUT PASSION'S RATION

DEFINITION OF KEYPOINTS

DEFINITION OF LINES DEFINITION OF MESH SIZE MESH ALL THE LINES

APPLY THE BOUNDARY CONDITIONS

APPLY THE VERTICAL FORCE

NSEL,S,LOC,X,Ø,5*L1 NSEL,R,LOC,Y,Ø F,ALL,FY,P NSEL,ALL ANTYPE,Ø /SOLU SOLVE FINISH ANPUT MACRO4

IDEFINE STATIC ANALYSIS

A.3.3 A SIMPLY SUPPORTED BEAM WITHOUT ROLLER SUPPORT SUBJECT TO A CONCENTRATED FORCE IN THE MIDDLE

LARGE DEFLECTION ANALYSIS USING MODIFIED ROBUST SECANT METHOD

/PREP7 *SET,EM,200E3

*SET,A1, 0.01 *SET,I1, 8.33e-6 *SET,H1,0.1 *SET,L1,1 *SET,LZ, 1/50 *SET,P.(32)*1e2

ET,1BEAM3 R,1,A1,11,H1 MP,EX,1,200e3 MP.NUXY.1.0

K,1,0,0 K,2,L1,0 L,1,2 LESIZE,ALL, LZ LMESH,ALL

NSEL,S,LOC,X,0 NSEL,R,LOC,Y,0 D,ALL,UX D.ALL,UY

NSEL,S,LOC,X,L1 NSEL,R,LOC,Y,0 D,ALL,UY D,ALL,UY

NSEL,S,LOC,X,0,5*L1 NSEL,R,LOC,Y,0 F,ALL,FY,-P NSEL,ALL ANTYPE,0 /SOLU SOLVE FINISH ANPUT MACRO4

JUSE BEAM 3 LEMENT

AREA MOMEMINT OF INITIAL IN MM⁴ HEIGH LENGTH MESH SIZE FORCE

USE BEAM 3 ELEEMNT INPUT GEOMETRICAL PROPERTIES INPUT YOUNG'S MODULUS INPUT PASSION'S RATION

DEFINITION OF KEYPOINTS

DEFINITION OF LINES DEFINITION OF MESH SIZE MESH ALL THE LINES

APPLY THE BOUNDARY CONDITIONS

APPLY THE VERTICAL FORCE

DEFINE STATIC ANALYSIS

APPENDIX B

B.1 MACRO1 FOR MODIFYING SECANT RIGIDITY IN PLASTIC ANALYSIS

*GET,SZ,ELEM,0,COUNT

/POST1

*DIM,COL1,ARRAY,SZ,1 *DIM,COL2,ARRAY,SZ,1 *DIM,COL3,ARRAY,SZ,1 *DIM,COL4,ARRAY,SZ,1 *DIM,COL4,ARRAY,SZ,1 *DIM,COL5,ARRAY,SZ,1 *DIM,COL7,ARRAY,SZ,1 *DIM,COL5,ARRAY,J,1

FTABLE MI SMISC 6

DEFINE THE NAMES OF

OBTAINING THE

1 DEFINE ARRAY PARAMETERS

ETABLE, MJ, SMISC, 12

*DO,KK,LSZ			
*GET.MI.ELEM.KK.ETAB.MI	GETTING T	HE	
	BENDING M	IOMENT OF	
*GET,MJ,ELEM,KK,ETAB,MJ	EACH ELES	4ENT	
VFILL,COL1(KK),DATA,MI	PUTTING:	THEM	INTO
PARRAYS			
*VFILL.COL2(KK),DATA,MJ	PUTTING	THEM	INTO
PARRAYS			
*SET.COL3(KK).(COL1(KK)+COL	2(KK))/2		
*ENDDO			
*VSCFUN,COL4(1),MAX,COL1(1)	SETTING THE M/	XIMUM BE	NDING
MOD	FYING THE PROPE	TIES	

*CFOPEN, MODIFY1 *DIM, IMODIFY, ARRAY, SZ, 1 *DIM, HMODIFY, ARRAY, SZ, 1 *DO, JJ, 1, SZ, 1 *GET MI FLEM, JJ, FTAR, MI *GET,MJ,ELEM,JJ,ETAB,MJ *VFILL,COL5(JJ),DATA,MI *VFILL,COL6(JJ),DATA,MJ *SET,COL7(JJ),(COL5(JJ)+COL6(JJ))/2

*SET,IMODIFY(JJ),(abs(COL4(J))COL7,(JJ))*1 MOMENT OF INITIAL *SET,JIMODIFY(JJ),(12*IMODIFY(JJ),(3**(1/2) MODIFY HEIGHT *CFWRITE,R,JJ,A,IMODIFY(JJ),JIMODIFY(JJ) INDIFIED #DEPETTS

*CFWRITE,REAL,IJ *CFWRITE,EMODIF,IJ *ENDDO *CFCLOS

"""""REANALYZE THE MODLE

/PREP7 RESUME *USE,MODIFY1 FINISH /SOLU SAVE SOLVE

B.2 MACRO2 FOR MODIFYING SECANT RIGIDITY IN

BUCKLING ANALYSIS

*GET.SZ.ELEM.@.COUNT

OBTAINING THE NUMBER OF ELEMENT

/POST1 *DIM,COL1,ARRAY,SZ,1 *DIM,COL2,ARRAY,SZ,1 *DIM,COL3,ARRAY,SZ,1

DEFINE ARRAY PARAMETERS

ETABLE,UX,U,X

DEFINE THE NAME OF LATERAL DISPLACEMENT

*DO,KK,1,SZ *GET,UX,ELEM,KK,ETAB,UX *VFILL,COL1(KK),DATA,UX *ENDDO *VSCFUN.COL2 (D,MAX.COL1(1)

GETTING THE MAXIMUM

IDISPLACEMENT

MODIFYING THE PROPETIES

*CFOPEN,MODIFY1 *DIM,IMODIFY,ARRAY,SZ,1 *DIM,IIMODIFY,ARRAY,SZ,1 *DO,IJ,1,SZ,1 *GET UY FLEM LI FTAB UY

*VFILL.COL3 (JJ),DATA,UX

*SET,IMODIFY(JJ),(abs(COL2 (1)/COL3 (JJ)))*1 *MODIFY MOMENT OF ! INERTIA

*SET,HMODIFY(JJ),(12*IMODIFY(JJ)/A)**(1/2) !MODIFY HEIGHT

*CFWRITE,R,JJ,A,IMODIFY(JJ),HMODIFY(JJ) INPUT MODIFIED PROPERTIES

*CFWRITE,REAL,JJ *CFWRITE,EMODIF,JJ *ENDDO

*CECLOS

/PREP7 RESUME *USE,MODIFY1 FINISH /SOLU SAVE NSEL,ALL SOLVE

B.3 MACRO3 FOR MODIFYING SECANT RIGIDITY IN

LARGE DEFLECTION ANALYSIS

ANALYSIS FOR CANTILEVER BEAMS

*GET.SZ.ELEM.0.COUNT	OBTAINING	THE	NUMBER	OF
ELEMENT				
/POST1				
*DIM.COL1.ARRAY.SZ.1	! DEFINE ARR	AY PAR	AMETERS	
ETABLE,UY,U,Y	DEFINE THE	NAME (OF	
	LATERAL DIS	PLACE	MENT	

*DO,KK,LSZ

*GET,UY,ELEM,KK,ETAB,UY *VFILL,COL1(KK),DATA,UY *ENDDO

*CFOPEN,MODIFY1 *DIM,IMODIFY,ARRAY,SZ,1 *DIM,HMODIFY,ARRAY,SZ,1

*DO,JJ,1,SZ,1 *GET,UY,ELEM,JJ,ETAB,UY *VFILL,COL1 (JJ),DATA,UY

*SET.IMODIFY(JJ),1.05*11

MODIFY MOMENT OF INITIAL MODIFICATION FACTOR IS 1.05

*SETJIMODIFY(JJ),(J2*IMODIFY(JJ),(J)**(IZ) *MODIFY (HEIGHT *CFWRITE,R,JJ,ALJMODIFY(JJ),JMODIFY(JJ) *CFWRITE,RRAL,JJ *CFWRITE,RMODIF,JJ *ENDDO

*CFCLOS

REANALYZE THE MODLE

/PREP7 RESUME *USE,MODIFY1 FINISH /SOLU SOLVE

B.4 MACRO4 FOR MODIFYING SECANT RIGIDITY IN

LARGE DEFLECTION ANALYSIS

I ANALYSIS FOR SIMPLY SUPPORTED BEAMS 'CETSZ-BLEMA,COUNT 'OBTAINING THE NUMBER OF ELEMENT 'POSTI 'DINCOLLARRAYSZI 'ETABLE,UY,UY 'DEFINE THE NAME OF 'LATERA'DISPLACEMENT

*DO.KK.1.SZ

*GET,UY,ELEM,KK,ETAB,UY *VFILL,COLI(KK),DATA,UY *ENDDO

MODIFYING THE PROPETIES OF THE BEAM

*CFOPEN,MODIFY1 *DIM,IMODIFY,ARRAY,SZ,1 *DIM,HMODIFY,ARRAY,SZ,1 *DO,JJ,I,SZ,1 *GET,UY,ELEM,JJ,ETAB,UY *VFILL_COL1 (JD,DATA,UY

*SET.IMODIFY(JJ).0.95*11

MODIFY MOMENT OF INITIAL MODIFICATION FACTOR IS 0.95

*SET,HMODIFY(JJ)(12*IMODIFY(JJ)(A1)**(L2) *MODIFY HEIGHT *CFWRITE,R,JJ,ALI,MODIFY(JJ)JIMODIFY(JJ) *CFWRITE,REMODIF,JJ *CFWRITE,REMODIF,JJ *NDDO

*CFCLOS

REANALYZE THE MODLE

/PREP7 RESUME *USE_MODIFY1 FINISH /SOLU SOLVE

APPENDIX C

C.1.1 A UNIFORM PORTAL FRAME SUBJECT TO A LATERAL

FORCE AND A VERTICAL FORCE

PLASTIC ANALSIS USING GEOMETICAL NONLINEAR METHOD PREP7 : ENTER PREPORCESSOR

ET,I,BEAM 23

'USE BEAM23 ELEMENT 'YOUNG'S MODULUS 'AREA IN MM² ! MOMENT OF INITIAL IN MM⁴ ! HEIGHT IN MM

1 MESH SIZE IN MM

*SET,EM,200E3 *SET,A,50 *SET,J, 266.67 *SET,H,8 *SET,LZ, 2 *SET,P1,8c2 *SET,P2, 7.2c2

R,1,A,I, H MP,EX,1, 200e3

TB, MISO

TBPT,,210 /(200e3),210 TBPT,,1.44E-03,230 TBPT,,2.08E-03,250 TBPT,,3.38E-03,270 TBPT,,7.25 E-03,290 TBPT,,1.5E-02,300 TBPT, 1.5E-02,300 ! LATERAL FORCE ! VERTICAL FORCE ! INPUT AREA,MOMENT INITIAL UNPUT YOUNG'S MODULUS

DEFINE MUTILINEAR MATERIAL

MP. NUXY J.0

K,1,0,0 K,2,0,400 K,3,400,400 K,4,400,0

L,1,2 L,2,3

DEFINE PASSION'S RATIO

: DEFINITION OF KEYPOINTS

DEFINITION OF LINES

LESIZE, ALL, 10 LMESH, ALL

NSEL,S,LOC,X,200 EALL-FY-P1 DEFINITON OF MESH SIZE

APPLY THE LATERAL FORCE

1 APPLY THE VERTICAL FORCE

NSEL,S,LOC,X,0 NSEL,R,LOC,Y,400 F,ALL,FX,P2

APPLY BOUNDARY CONDITION

NSEL,S,LOC,y,0 D,ALL,ALL NSEL,ALL

/SOLU ANTYPE,0 OUTRES,ALL,ALL

DEFINE STATIC ANALYSIS

DELTIM,0.01,0.001,0.01 LNSRCH,ON NCNV, 1 SOLVE DEFINE TIME SIZE STEP USE LNSRCH TEHCNIQUE

C.1.2 A TWO BAY AND ONE STOREY FRAME SUBJECT TO

TWO VERTICAL FORCES AND ONE LATERAL FORCE

PLASTIC ANALSIS USING GEOMETICAL NONLINEAR METHOD (PREP7 : ENTER PREPORCESSOR

ET.I.BEAM 23

!USE BEAM23 ELEMENT

*SET,EM,200E3 *SET,A,50 *SET,I,266.67 *SET,I,8 *SET,LZ,10 *SET,P1,6e2 *SET,P2,7.2e2 *SET,P3,5e2

R,1,A,I, H MP,EX,1, 200e3 TB, MISO

TBPT,,210 /(200e3),210 TBPT,,1.44E-03,230 YOUNG'S MODULUS IN SAREA MOMENT OF INERTIA HEIGHT MESH SIZE VERTICAL FORCE VERTICAL FORCE LATERAL FORCE

INPUT AREA, MOMENT INITIAL INPUT YOUNG'S MODULUS DEFINE MUTILINEAR MATERIAL IDEFINE DIFFERENT STRESS AND STRAIN POINTS

TBPT.,2.08E-03,250
TBPT,,3.38E-03,270
TBPT,,7.25 E-03,290
TBPT,,1.5E-02,300
TBPT., 3.0E-02 .300

MP, NUXY J.0

K,1,0,0 K,2,0,400 K,3,400,400 K,4,400,0 K,5,800,400 K,6,800,0

L,1,2 L,2,3 L,3,4 L,3,5

LESIZE, ALL, 10 LMESH, ALL

NSEL,S,LOC,X,200 FALL,FY,-P1

NSEL,S,LOC,X,600 FALL,FY,-P2

NSEL,S,LOC,X,800 NSEL,R,LOC,Y,400 F,ALL,Fx,P3 DEFINE PASSION'S RATIO

DEFINITION OF KEYPOINTS

DEFINITON OF MESH SIZE MESH LINES

IDEEINITION OF LINES

APPLY THE VERTICAL FORCE

APPLY THE VERTICAL FORCE

1APPLY THE LATERAL FORCE

APPLY BOUNDARY CONDITION

NSEL,S,LOC,y,0 D,ALL,ALL NSEL,ALL

/SOLU

ANTYPE,0 OUTRES,ALL,ALL DELTIM,0.01,0.001,0.01 LNSRCH,ON NCNV, 1 SOLVE DEFINE STATIC ANALYSIS

DEFINE TIME SIZE STEP USE LNSRCH TEHCNIQUE

C.1.3 A UNIFORM FRAME SUBJECT THREE VERTICAL FORCES AND ONE LATERAL FORCE

PLASTIC ANALSIS USING GEOMETICAL NONLINEAR METHOD

/PREP7

! ENTER PREPORCESSOR

ET,I,BEAM 23

TISE REAM23 ELEMENT

*SET,EM,200E3 *SET,A,50 *SET,I,6.25*8*8*8/12 *SET,H,8 *SET,L,150 *SET,LZ,2 *SET,P1,4e3 *SET,P2,2.5e3

R,1, A,I, H MP,EX,1, 200e3 TB, MISO

TBPT,,210 /(200e3),210 TBPT,,1.44E-03,230 TBPT,,2.08E-03,250 TBPT,,3.38E-03,270 TBPT,,7.25 E-03,290 TBPT,,1.5E-02,300 TBPT,, 3.0E-02,300

MR NUVY 1.6

YOUNG'S MODULUS 'AREA ' MOMENT OF INITIAL ' HEIGHT ' LENGTH ' MESH SIZE ' FORCES

INPUT AREA, MOMENT INITIAL INPUT YOUNG'S MODULUS DEFINE MUTILINEAR MATERIAL DEFINE DIFFERENT STRESS AND STRAIN POINTS

DEFINE PASSION'S RATIO

R.1.99,0
K.2.0.L
K,3,0.8°L,I
K.4.0.8*L.0
K,5,0,2*L
K.6.0.8*L.2
K.7.1.3*L.I
K.8.1.3*L.0
L.1.2
1.2.3
1.3.4
1.0.0

DEFINITION OF LINES

L,5,6 L,6,3

L,3,7 L,7,8

LESIZE, ALL, LZ LMESH, ALL /SOLU

NSEL,S,LOC,X,0.4*L NSEL,R,LOC,Y,L F,ALL,FY,-P1

NSEL,S,LOC,X,0.4*L NSEL,R,LOC,Y,2*L FALL,FY,-P1

NSEL,S,LOC,X,1.05*L NSEL,R,LOC,Y,L F,ALL,FY.-P2

NSEL,S,LOC,X,0 NSEL,R,LOC,Y,2*L FALL,Fx,P2 DEFINITON OF MESH SIZE

APPLY THE VERTICAL FORCE

APPLY THE LATERAL FORCE

NSELALL

NCNV, 1

APPLY BOUNDARY CONDITION

NSEL,S,LOC,y,0 D,ALL,ALL NSEL,ALL /SOLU ANTYPE,0 OUTRESALLALL

DELTIM,0.01,0.001,0.01 LNSRCH ON DEFINE STATIC ANALYSIS

DEFINE TIME SIZE STEP USE LNSRCH TEHCNIOUE

C.I.4 A UNIFORM BIG FRAME SUBJECT TO UNIFORMLY LATERAL DISTRIBUTED FORCES AND VERTICAL CONCENTRATED FORCES

PLASTIC ANALSIS USING GEOMETICAL NONLINEAR METHOD

/PREP7

ET.I.BEAM 23

*SET,EM,200E3 *SET,A,50 *SET,1,6.25*8*8*8/12 *SET,H,8 *SET,H,150 *SET,LZ,2 *SET,P1,1c2 *SET,P2,1.5c2 *SET,P3,1c2

R,1, A,I, H MP, EX,1, 200e3 TB, MISO

TBPT,,210 /(200e3),210 TBPT,,1.44E-03,230 TBPT,,2.08E-03,250 TBPT,,3.38E-03,250 TBPT,,7.25 E-03,290 TBPT,,1.5E-02,300 TBPT,, 3.08:-02,300

MP, NUXY ,1,0 K,1,0,0 K,2,1,2*L,0 K,3,2,4*L,0 K,4,3,4*L,0

K,6,0,L K,7,1.2*L,L K,8,2.4*L,L K,9,3.4*L,L K,19,4,4*L,L

K,11,0,2*L K,12,1.2*L,2*L K,13,2.4*L,2*L K,14,3.4*L,2*L K,15,4.4*L,2*L

K,16,1.2*L,3*L K,17,2.4*L,3*L K,18,3.4*L,3*L

! ENTER PREPORCESSOR

USE BEAM23 ELEMENT

YOUNG'S MODULUS 'AREA IN MM² 'MOMENT OF INITIAL 'HEIGHT 'LENGTH 'MESH SIZE 'FORCES 'FORCES 'FORCES

: INPUT AREA, MOMENT INITIAL INPUT YOUNG'S MODULUS DEFINE MUTILINEAR MATERIAL DEFINE DIFFERENT STRESS AND STRAIN POINTS

DEFINE PASSION'S RATIO DEFINITION OF KEYPOINTS

K.19.1.2*L.4*L	
K.20.2.4*L.4*L	
K.21.3.4*L.4*L	
L,1,6	DEFINITION OF LINES
L,2,7	
L.3.8	
L,4,9	
L,5,10	
L,6,7	
L,7.8	
L,8,9	
L,9,10	
1.4.11	
1,6,11	
1,7,12	
1,6,13	
1,10,15	
1,10,15	
1,11,12	
1,12,13	
1,15,14	
1,14,15	
L.12.16	
L.13.17	
L,14,18	
L,16,17	
L,17,18	
1.15.10	
1,10,17	
1,17,20	
1,10,10	
1,17,40	
1.,20,21	

LESIZE,ALL,LZ LMESH,ALL /SOLU

ESEL,S,ELEM,,1,75 SFBEAM,ALL,1,PRES,P1

ESEL,S,ELEM,,706,780 SFBEAM,ALL,1,PRES,P1

APPLY THE LATERAL FORCE

ESEL,S,ELEM,,1411,1485 SFBEAM,ALL,1,PRES,P1

ESEL,S,ELEM, 1801,1875 SFBEAMALL, LPRES,P1

ESEL,S,ELEM,1336,1410 SFBEAM,ALL,1,PRES,P2

ESEL,ALL,ALL

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,L FALL,FY,-P3

APPLY THE VERTICAL FORCE

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,2*L F,ALL,FY,-P3

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,3*L F,ALL,FY,-P3

NSEL,S,LOC,X,1.8*L NSEL,R,LOC,Y,4*L F,ALL,FY,-P3

NSEL, SLOC, 3, J DALL, ALL NSEL, ALL ANTYPE, J OUTRES, ALL, ALL SAVE NSEL, ALL NOLU ANTYPE, J OUTRES, ALL, ALL DELTIM, 0, 0, 0, 0, 0, 0, 1 LNSRCH, (0) NCNV, 1 SOLVE

DEFINE STATIC ANALYSIS

DEFINE TIME SIZE STEP USE LNSRCH TEHCNIQUE

C.1.5 A NON-UNIFORM PORTAL FRAME SUBJECT TO A

LATERAL FORCE AND A VERITICAL FORCE

PLASTIC ANALSIS USING GEOMETICAL NONLINEAR METHOD

PREP7 ET. LBEAM23

*SET EM 200E3 *SET.L.800

*SET.A1.3000

1 ENTER PREPORCESSOR TISE REAM 22 ELEMENT

TVOLING'S MODULUS IN N/MM2

DEFINE GEOMETRICAL **PROPERTIES** DIFFERENT

*SET.I1.50*60*60*60/12 *SET.H1.60

*SET.A2.5000 *SET,12,50*100*100*100/12 *SET.H2.100

*SET,A3,7000 *SET.I3.50*140*140*140/12 *SET.H3.140

*SETLZ_5 *SET PL 1E6 *SET.P2.1E6

IMESH SIZE IN MM

INPUT AREA MOMENT INITIAL

INPUT YOUNG'S MODULUS DEFINE MUTILINEAR MATERIAL 'DEFINE DIFFERENT STRESS AND STRAIN POINTS

R.I.ALILH1 R 2 A2 12 H2 P 1 A 1 I H 1 MP.EX.1, 200e3 TR MISO

TBPT.,210 /(200e3),210 TRPT_1_44E_03.230 TBPT_2.08E-03.250 TBPT_3.38E-03.270 TRPT 7 25 E-03 290 TBPT_1.5E-02.300 TRPT., 3.0E-02.,300

MP. NUXY J.0 K.1.0.0 K.2.0.L K.3.3/2*L.L. K 4 3/2*L 0 L.3,4

LSEL,S.,I

'DEFINE PASSION'S RATIO 1 DEFINITION OF KEYPOINTS

IDEFINITION OF LINES

ISELECT LINE 1

180

LATT_2 LINE 1 LSEL_S_2 LATT_1 LINE 2 LSEL_S_3 LATT_3 LINE 3 LSEL_ALL LSEL_F ALL 17

I MESH ALL

APPLY REALCONSTANT 2 TO

SELECT LINE 2 APPLY REALCONSTANT 1 TO

SELECT LINE 3 APPLY REALCONSTANT 3 TO

DEFINITON OF MESH SIZE

! MESH LINES

APPLY THE LATERAL FORCE

NSEL,S,LOC,X,Ø NSEL,R,LOC,Y,L F,ALL,FX,P1 NSEL,S,LOC,X,3/4*L NSEL,R,LOC,Y,L F,ALL,FV,-P2

NSEL,S,LOC,y,0 CONDITION APPLY THE VERTICAL FORCE

: APPLY THE BOUNDARY

D,ALL,ALL SAVE NSEL,ALL /SOLU ANTYPE,0 OUTRES ALL ALL

LNSRCHON

NCNV, 1 SOLVE

DELTIM.0.01.0.001.0.01

: DEFINE STATIC ANALYSIS

DEFINE TIME SIZE STEP USE LNSRCH TEHCNIOUE

C.2.1 A UNIFORM PORTAL FRAME SUBJECT TO THREE

VERTICAL FORCES ON THE BEAM

BUCLING ANALSIS USING GEOMETICAL NONLINEAR METHOD

/PREP7 ET,1,BEAM23

JUSE BEAM 23 LEMENT

*SET,EM,200E3 *SET,L1, 600 YOUNG'S MODULUS

*SET.L.2, 800

*SET,A, 80 *SET, I, 8*10*10*10/12 *SET,H, 10 *SET,LZ, 5 *SET,P1,2.0E4 *SET,P2,1.5E4 *SET,P2,1.2E4

R,1,A,I,H INITIAL UIMP,I, EX,,,EM MP,NUXY,1,0

k,1,0,0 k,2,0,L1 k,3,L2,L1 K,4,L2,0

L,1,2 L,2,3

LESIZE,ALL,LZ LMESH,ALL /SOLU

NSELS LOC.Y.0

D.ALL.ALL

NSEL,S,LOC, X, 0.3 *L2 FORCES NSEL,R,LOC, Y,L1 FALL,FY,-P1

NSEL,S,LOC,X,0.5*L2 NSEL,R,LOC,Y,L1 FALL,FY,-P2

NSEL,S,LOC,X,0.8 *L2 NSEL,R,LOC,Y,L1 F,ALL,FY,-P3

NSEL, ALL ANTYPE,0 OUTRES, ALL, ALL NLGEOM, ON CALCULATION AREA MOMENT OF INITIAL HEIGHT MESH SIZE FORCE IN NEWTON

INPUT AREA, MOMENT

INPUT YOUNG'S MODULUS INPUT PASSION'S RATION

DEFINITION OF KEYPOINTS

IDEFINITION OF LINES

1 MESH LINES

APPLY THE BOUNDARY CONDITIONS

: APPLY THE VERTICAL

IDEFINE STATIC ANALYSIS

LARGE

DEFLECTION

DELTIM, 0.001,0.0001,0.001 SOLVE

DELTIM, 0.001,0.0001,0.001 DEFINE TIME SIZE STEP

JUSE REAM 23 LEMENT

C.2.2 A UNIFORM PORTAL FRAME SUBJECT TO A VERTICAL

FORCE AND A LATERAL FORCE

BUCLING ANALSIS USING GEOMETICAL NONLINEAR METHOD

/PREP7 ET.I.BEAM23

YOUNG'S MODULUS IN N/MM²

*SET,EM,200E3 *SET,L1,600 *SET,L2,800

*SET,A, \$0 *SET, I, 8*10*10*10/12 *SET,H, 10 *SET,L,Z, 5 *SET,P1,2.0E4 *SET,P2,1.5E4 *SET,P2,1.5E4

UIMP.1. EX., EM

AREA IN MM² MOMENT OF INITIAL IN MM⁴ HEIGHT IN MM MESH SIZE IN MM FORCE IN NEWTON

1 INPUT AREA, MOMENT INITIAL 21NPUT YOUNG'S MODULUS 21NPUT PASSION'S RATION

DEFINITION OF KEYPOINTS

MP,NUXY,1,0 k,1,0,0 k,2,0,L1 k,3,L2,L1 K,4,L2,0

L,1,2 L,2,3 L,34

LESIZE,ALL,LZ LMESH,ALL /SOLU

NSEL,S ,LOC,Y ,0 D,ALL,ALL

NSEL,S,LOC, X, 0.5 *L2 NSEL,R,LOC, Y,L1 F,ALL,FY,-P1 DEFINITION OF LINES

1 MESH LINES

1 APPLY THE BOUNDARY CONDITIONS

1 APPLY THE VERTICAL FORCES

NSELS LOC X 0 NSEL.R.LOC.Y.L1 EALL.FY.P2 NSEL, ALL

1 DEFINE STATIC ANALYSIS

ANTYPE.0 OUTRES, ALL, ALL NLGEOM, ON DELTIM 0.001.0.0001.0.001

! LARGE DEFLECTION CALCULATION DEFINE TIME SIZE STEP

C.3.1 A CANTILEVER BEAM SUBJECT TO A CONCENTRATED

FORCE AT THE FREE END

LARGE DEFLECTION ANALYSIS USING GEOMETRICAL NONLINEAR ANALYSIS PREPT *SET.EM.200F1 TISE REAM 31 EMENT

*SET.A, 8550 *SET.I. 1.04E+08 *SET H 257 *SET.L1.6000 *SET.L.Z. L1/50 *SET P(32)*1e2

ET.I.REAM 3 RIATH MP.EX.1.200e3 MP.NUXY.1.0

K.1.0.0 K2110

LESIZE ALL LZ LMESH,ALL

NSELS LOC X.0 D.ALL.ALL

NSELSLOCX, L1 NSEL.R.LOC.Y.0 FALL FY.P NSEL ALL

PAREA IN MM2 MOMEMNT OF INITIAL IN MM4 THEIGH IN MM TENCTH IN MM 1 MESH SIZE IN MM FORCE IN NEWTON

JUSE REAM3 ELEMENT **UNPUT GEOMETRICAL PROPERTIES** UNPUT VOUNC'S MODULUS INPUT PASSION'S RATIO

IDEEINITION OF KEVPOINTS

'DEFINITION OF LINES

IMESH LINES

1APPLY THE BOUNDARY CONDITIONS

APPLY THE VERTICAL LOAD

SOLU ANTVPE.0 OUTRES ALL ALL NICEOM ON DELTIM. 0.001.0.0001.0.001 SOLVE

INFEINE STATIC ANALYSIS

31 ARCE DEELECTION CALCULATION IDEEINE TIME SIZE STER

C.3.2 A SIMPLY SUPPORTED BEAM WITH ONE ROLLER SUPPORT SURJECT TO A CONCENTRATED FORCE IN THE MIDDI F

LARGE DEFLECTION ANALYSIS USING GEOMETRICAL NONLINEAR ANALYSIS PREP7 *SETEM 200E3 TISE REAM 3 LEMENT

*SET A. 8550 *SET1 104E+08 *SET.H.257 *SET L1 6000 ASETLZ LUM *SET.P.(32)*1#2

ET I DE AM 3 RIALH MPEX 1 200+1

K100

LESIZE ALL LZ I MESHALL

NSELSLOC X.0 NSEL RLOC VO DALL UX D.ALL-UY

NSEL STOCKT1 NSEL.R.LOC.Y.0 DALL UV

NEEL SLOC VARIA

IAREA IN MM² MOMEMNT OF INITIAL IN MM THEICH IN MM TENGTH IN MM 1 MESH SIZE IN MM TECHCE IN NEWTON

TIME REAMS ELEMENT INPUT GEOMETRICAL PROPERTIES UNPUT VOUNC'S MODULUS UNPUT PASSION'S PATIO

IDEFINITION OF KEYPOINTS

IDEFINITION OF LINES.

IMPOUTINES.

14 PPLY THE BOUNDARY CONDITIONS

APPLY THE VERTICAL FORCE

NSEL,R,LOC,Y,0 F,ALL,FY,-P NSEL,ALL

/SOLU

ANTYPE,0 OUTRES, ALL, ALL NLGEOM, ON DELTIM, 0.001,0.0001,0.001 SOLVE

! DEFINE STATIC ANALYSIS

LARGE DEFLECTION CALCULATION DEFINE TIME SIZE STEP

C.3.3 A SIMPLY SUPPORTED BEAM WITHOUT ROLLER SUPPORT SUBJECT TO A CONCENTRATED FORCE IN THE MUDDLE

 ! LARGE DEFLECTION ANALYSIS USING GEOMETRICAL NONLINEAR ANALYSIS

 PRRP1

 "SET_EA_300E3

 "USE BEAM 3 LEMENT

HEIGH IN MM

MESH SIZE IN MM

TISE REAMS ELEMENT

UNPUT VOUNC'S MODULUS

UNPUT PASSION'S PATIO

*SET,A, 8550 *SET,I, 1.04E+08 *SET,H,257 *SET,L1,6000 *SET,LZ, L1/50 *SET,P,(32)*1e2

ET,1,BEAM 3 R,1,A,I,H MP,EX,1,200e3 MP,NUXY,1,0

K,1,0,0 K,2,L1,0 L,1,2

LESIZE,ALL,LZ LMESH,ALL

NSEL,S,LOC,X,Ø NSEL,R,LOC,Y,Ø D,ALL,UX D.ALL,UY DEFINITION OF KEYPOINTS DEFINITION OF LINES MESH LINES

1 MOMEMNT OF INITIAL IN MM

INPUT CEOMETRICAL PROPERTIES

APPLY THE BOUNDARY CONDTIONS

NSELSLOCX.L1

NSEL,R,LOC,Y,0 D,ALL,UY D,ALL,UY

NSEL,S,LOC,X,0.5*L1 NSEL,R,LOC,Y,0 F,ALL,FY,-P NSEL.ALL **APPLY THE VERTICAL FORCE**

/SOLU

ANTYPE,0 OUTRES, ALL, ALL NLGEOM, ON DELTIM, 0.001,0.0001,0.001 SOLVE

DEFINE STATIC ANALYSIS

187

LARGE DEFLECTION CALCULATION DEFINE TIME SIZE STEP







